

European Journal of Physical Education and Sport Science

ISSN: 2501 - 1235 ISSN-L: 2501 - 1235 Available on-line at: <u>www.oapub.org/edu</u>

doi: 10.5281/zenodo.3864522

Volume 6 | Issue 4 | 2020

EFFICACY OF PHYSICAL ACTIVITY COUNSELLING INTERVENTIONS DELIVERED IN PRIMARY CARE: A SYSTEMATIC REVIEW AND META-ANALYSIS

Micky Olutende Oloo¹ⁱ, Edwin Kadima Wamukoya¹, Maximilla Wanzala² ¹Department of Health Promotion and Sports Science, Masinde Muliro University of Science and Technology, Kenya ²Department of Public Health, Masinde Muliro University of Science and Technology,

Kenya

Abstract:

Background: Overwhelming success has been achieved in disease control through environmental interventions such as vaccinations and improved hygiene to increase life expectancy, many authorities in the field of preventative healthcare are of the opinion that too little has been done to target behavioral factors, particularly physical inactivity. Research suggests that the impact PA counselling in primary care could have at a population level is tremendous. However, gaps still exist between the science and the practice when it comes to prescribing exercise in the healthcare context. Objectives: To compare the effectiveness of health provider delivered interventions for physical activity (PA) promotion verses placebos or no or minimal intervention among community dwelling adults Search methods: We searched Cochrane Central Register of Controlled Trials in the Cochrane Library, Ovid MEDLINE(R), Embase Ovid, Web of Science, CINAHL (EBSCOhost) and SPORTDiscus (earlier dates to 10 January, 2020) electronic databases regardless of language or publication status using the optimal sensitive search strategy developed by The Cochrane Collaboration. We used Medical Subject Headings (MeSH) [1], Looked up words in text word, abstract, title [2], then Combined [1] with [2] using Boolean logic (OR) and then Set up proper filters. Selection criteria: Randomised controlled trials (RCTs) and cluster randomised trials. We excluded quasi-RCTs and cross-over trials. We will include studies comparing health provider counselling intervention to placebo or no counselling/exercise prescription. We excluded studies that had more than a 20% loss to follow-up if they did not apply an intention-to treat analysis. The studies were considered if the outcomes were measured on a continuous scale and

ⁱ Correspondence: email <u>micky.oloo.mf@gmail.com</u>

results reported in terms on mean change, confidence intervals of change, standard deviations or standard error and mean. Studies with dichotomous outcomes were excluded. Data collection and analysis: Two review authors (EKW and MW) independently carried out data extraction for each included record using a pro forma specifically designed for the purpose. We resolved differences in data extraction by discussion. We plotted the results of each trial as point estimates, using means and standard deviations for the continuous outcomes. Since studies reported different outcome measures but measured the same concept, we calculated the standardised mean difference (SMD) with 95% confidence interval (CI) using a random-effects models. Main results: There were 15,269 apparently healthy adults who participated in the 24 included studies. All studies recruited both genders. The stated age range of participants was from 18 to 80 years. Meta-analysis of data from these trials suggests that health provider-led physical activity counselling interventions in primary care may lead to increased selfreport physical activity (SMD 0.11, 95% CI 0.04 to 0.17; participants = 13211; studies = 16; $I^2 = 50\%$), total energy expenditure (SMD 0.20, 95% CI 0.13 to 0.27; participants = 3376; studies = 8; $I^2 = 2\%$; overall effect P=0.00001) and systolic blood pressure at six showed a mean difference (MD) favouring health provider-led physical activity counselling interventions compared to usual care of -0.10 mmHg (SMD 0.27 95% CI 0.72 to 0.18;I² = 0%, overall effect P = 0.006) among patients. However meta-analysis showed that health provider-led physical activity counselling interventions in primary care did not lead to increased aerobic fitness (SMD 0.06, 95% CI -0.01 to 0.12; I² = 35%), body mass index (SMD -0.04, 95% CI -0.15 to 0.07) and total cholesterol (SMD -0.27, 95% CI -0.72 to 0.18 and Waist circumference (SMD -0.05, 95% CI-0.15 to 0.06; I² = 0%;overall effect P=0.24). Although there were limited data, there was no evidence of an increased risk of adverse events. Authors' conclusions: Counselling delivered by health providers, probably leads to similar or better physical activity outcomes for patient conditions (moderate-certainty evidence). However, these results must be interpreted with a degree of caution, recognising the variation in interventions reported within studies and the complex interplay of factors affecting outcomes. Several studies included multiple intervention methods, which made it difficult to tease apart which intervention components were the active ingredients

Keywords: physical activity counselling; interventions; primary care

1. Introduction

1.1 Plain language summary

1.1.1 Physical activity counselling interventions for promoting physical activity

In order to initiate important increases in physical activity, both exercise prescription and counselling should be incorporated in standard practice. We recommend as a tool for primary health care physicians to promote PA, especially at health check and control visits, where more time can be allocated for the appointment. A more comprehensive

familiarization protocol or training sessions are needed to implement physical counselling to everyday practices and facilitate counselling cooperation. Self-monitoring of PA with expert feedback can be a useful and cheaper way of increasing especially the duration of overall weekly PA in the short term.

1.2 Background

Physical activity (PA) improves quality of life and health in clinical and non-clinical populations, including prevention and management of non-communicable diseases (NCDs) (Tero et al., 2017; Lee et al.m 2012) like diabetes (Thent, Das & Henry, 2013), lowers blood pressure and reduces the risk of coronary heart disease, hypertension and stroke (Soares-Miranda et al., 2016; Green et al., 2011), reduces risk of developing breast and colon cancer (Kimmel Haas & Hermanns 2014) and has beneficial effects on body weight, fat mass and central obesity (Wiklund, 2016). A study by Machio, 2012 showed that NCDs reduced labour force participation by 61% but with elimination of physical inactivity, life expectancy in Kenya was expected to increase by between 0.25-0.49 years (Lee et al., 2012). Regular PA can achieve parallel or greater effects on NCDs' risk factors than those achieved with drugs at a lower cost and with minimal adverse effects (Fiuza-Luces et al., 2013). In a landmark British Medical Journal paper examining the head-tohead effects of medication versus PA/exercise in chronic disease, Naci & Ioannidis, 2013 from Stanford University made a strong case for equivalent or superior effect of the health benefits of PA. In particular, PA interventions were more effective than drug treatment among patients with stroke and were as effective as medications for the prevention of diabetes and secondary treatment of NCDs (Naci & Ioannidis, 2013).

Research suggests that the impact PA counselling in primary care could have at a population level is tremendous. Goldstein et al., 1999 noted that the interaction between the high prevalence of sedentary behavior and the frequency of physician visits, coupled with primary care PA promotion had the potential to significantly impact the incidence of hypokinetic diseases such as heart disease, stroke, hypertension. In 2005, the American College of Preventive Medicine issued a position statement "that primary care providers should incorporate PA counselling into routine patient visits" (Jacobson other professional organizations echo the American Academy of Family Physicians, the American Academy of Paediatrics, the American College of Obstetrics and Gynaecology, the American Heart Association, National Institutes of Health, and the Surgeon General (Garry, Diamond & Whitley, 2002; Jacobson et al., 2005). The ACSM is yet another organization that recognizes and endorses the importance of primary care PA counselling through its initiative, "Exercise is Medicine" (Sallis, 2011). This initiative seeks to create awareness that "exercise is medicine" and should be prescribed accordingly. Previous Cochrane reviews found that PA interventions had a moderate effect on participation levels (Foster et al 2005; Richards, Hillsdon, Thorogood & Foster, 2013). This review seeks to give and overview of the effectiveness of all health provider delivered intervention in improving physical activity outcomes in adult populations. It also seeks to give more detail to a narrative review done by <u>Oloo, Kweyu & Ashiali, 2017</u>.

1.3 Description of the condition

It is evaluated that in 2008 physical inactivity caused 9% of untimely mortality and 5.3 million deaths around the world (Lee et al., 2012). This included between 6% to 10% of all deaths from major non-transferable maladies all-inclusive. It has been assessed that expanding the number of individuals that accomplish the WHO PA proposals by 10% or 25% would forestall more than 553,000 and 1.3 million deaths, separately, comprehensively every year (Lee et al., 2012). In Kenya, research shows that NCDs have been a growing problem over the years (WHO, 2017; Machio, 2012). In 2012 NCDs accounted for more than 50% of total hospital admissions and over 55% of hospital deaths in Kenya (Kenya Demographic and Health Survey, 2014). According to Kenya Demographic and Health Survey [KDHS] report (2014), over 61% of the population in Kenya did not engaged in exercise that caused an increase in their heart rate for at least 10 minutes continuously at work or during other activities. In the Western region of Kenya, results showed that 39.1% of women and 45.4% of men did not engage in PA at all (Grimstvedt et al., 2013). This corroborates with studies done in other parts of the world that showed most adults world- wide did not engage in PA at levels with the potential to yield benefits (Ding et al., 2016; Das & Horton, 2016; Hallal., 2012).

1.4 Description of the intervention

Studies have shown that HCPs have the best opportunity to question and counsel patients about the importance of PA since they can take advantage of the on-going care they provide to a large sector of the population and be influential in changing patients' behaviours. (Lobelo & de Quevedo, 2016; O'Brien et al., 2017; Lamming et al., 2017). Research has proven that HCPs have access to a large proportion of the sedentary population (Vuori et al., 2013; Matheson et al., 2011) and are therefore considered to be well positioned to champion the course of prevention of chronic diseases by prescribing PA during standard consultation (Teferi, Kumar & Singh, 2017; Matheson et al., 2011). In Western Kenya for example, Kakamega county which is the current study's setting is the second most populous county in Kenya with a nurse patient ratio of 86.37 per 100,000 people, which is higher than the national average of 51.5 per 100,000 people (HMIS, 2012). This is an indication that many HCPs in the county have access to a large proportion of the sedentary population.

1.5 How the intervention might work

Behavioural theories provide a foundation for an intervention that can explain the drivers of PA behaviour and potential pathways for change. It is now generally accepted that well designed PA interventions are based upon behavioral theories (<u>Bartholomew et al., 2001</u>). The majority of studies have adopted social psychology theories (<u>Biddle & Mutrie, 2001</u>). Some of the common theories used to explain behaviour include Social cognitive theory, Transtheoretical Model of behaviour change and Theory of planned behaviour. These theories have conceptual convergence and also share two common constructs (<u>Biddle & Mutrie, 2001</u>; <u>Rodgers & Brawley, 1991</u>). These constructs are the outcome

expectancy, which is the belief that the behaviour will lead to a specific outcome and social norms which refers to the influence of expected behaviour within a social group. With regards to PA behaviour, experts agree that variation in PA behaviour can be explained by factors operating at two levels, the intrinsic and extrinsic levels (Sallis, 2009). Intrinsic factors have received greater attention than the external factors in attempts to explain behavioral choices. Exercise prescriptions from HCPs will remind patients that PA is part of their treatment plan and should be adhered to with the same diligence with which their medication is taken (Grandes et al., 2009). Almost two-thirds of patients (65%) would be more interested in exercise and PA to stay healthy if advised by their HCP (Leemrijse et al., 2015), while 24% of patients will turn to fitness and health web sites for advice on exercise and PA but after consulting their HCP first (25%) (Derman et al., 2008). Researchers have confirmed that majority of people cite their HCP as their primary source of information regarding healthy lifestyle decisions (diet and exercise) (Lanhers et al., 2015; Leemrijse et al., 2015).

1.6 Why it is important to do this review

There is no "silver-bullet" to solve the global physical inactivity, an "all of the above" and "whole of society" approach, including PA counselling and referral in the health care context, will be required (Heath et al., 2012). While overwhelming success has been achieved in disease control through environmental interventions such as vaccinations and improved hygiene to increase life expectancy, many authorities in the field of preventative healthcare are of the opinion that too little has been done to target behavioral factors, particularly physical inactivity (Sallis, 2011). Once we have a better understanding of these factors, it will help better inform and target training efforts of current and future HCPs, and more importantly encourage HCPs to integrate exercise within standard consultation to further improve the health and well-being of clients. HCPs will begin to develop potentially better biopsychosocial treatment plans and interventions that include individually tailored exercise plans that help clients live increasingly holistic healthy lives. Understanding the effectiveness of these more traditional approaches to implementation should influence PA policy makers and professionals.

1.7 Objectives

To compare the effectiveness of health provider delivered interventions for physical activity (PA) promotion verses placebos or no or minimal intervention among community dwelling adults.

2. Methods

2.1 Criteria for considering studies for this review

2.1.1 Types of studies

Randomised controlled trials (RCTs) and cluster randomised trials. We excluded quasi-RCTs and cross-over trials. We will include studies comparing health provider counselling intervention to placebo or no counselling/exercise prescription. We excluded studies that had more than a 20% loss to follow-up if they did not apply an intention-to treat analysis. The studies were considered if the outcomes were measured on a continuous scale and results reported in terms on mean change, confidence intervals of change, standard deviations or standard error and mean. Studies with dichotomous outcomes were excluded.

2.1.2 Types of participants

Community dwelling participants men and women (18 years and above) randomly assigned. Studies that only assessed men or women alone were also included. Exclusion criteria included participants who had secondary hypertension, mental or physical illness serious enough to potentially influence the compliance with the study procedures, alcoholism, type 1 diabetes, current or planned pregnancy and history of myocardial infarction or stroke within the preceding 3 months.

2.1.3 Types of interventions

Physicians, general practitioners, nurses, nutritionist in a primary care setting. Relevant interventions include, but not limited to:

- counselling or advice, or both;
- self-directed or prescribed exercise, or both;
- home based, telephone based or facility-based exercise, or both;
- written education or motivational support material, or both.

Telephone counselling. Intervention and comparators are eligible regardless of delivery or duration.

2.1.4 Types of outcome measures

A. Primary outcomes

The primary outcomes of this review included data that assessed change between control and intervention for:

- self-report measures of PA behaviour. This was any physical activity outcome assessed by questionnaires (Total energy expenditure (Kcal/week or Kcal/day) and physical activity (minutes/week or hours/week))
- aerobic fitness (VO2max) (ml/kg/min or ml/min). It was either estimated from a submaximal fitness test or recorded directly from a maximal

B. Secondary outcomes

The secondary outcomes of this review included data for:

- 1. clinical anthropometric parameters. This was assessed by body mass index (BMI) and body circumferences (waist).
- 2. systolic blood pressure assessed in millimetres of Mercury (mmHg)
- 3. total cholesterol.

2.2 Search methods for identification of studies

2.2.1 Electronic searches

We searched the following electronic databases regardless of language or publication status using the optimal sensitive search strategy developed by The Cochrane Collaboration (Lefebvre, 2011). We used the PICO (P-patient problem, I-intervention, C-comparison, O-outcomes) acronym to help keep focus on the key concepts. We Used Medical Subject Headings (MeSH) [1], Looked up words in text word, abstract, title [2], then Combined [1] with [2] using Boolean logic (OR) and then Set up proper filters e.g. randomised controlled trials. We also borrowed the search strategies of a previous systematic review (Richards, Hillsdon, Thorogood & Foster, 2013). The searches were based on the MEDLINE search strategy combined with the sensitivity- and precision-maximising version of the Cochrane Highly Sensitive Search Strategy for identifying RCTs (Lefebvre, 2011). We modified the search strategy to use in the other databases. Details of the search strategy are presented in the Appendices (Appendices)

- 1) Cochrane Central Register of Controlled Trials (January, 2020) in the Cochrane Library;
- 2) Ovid MEDLINE(R) and In-Process & Other Non-Indexed Citations and Daily (1948 to 17 January, 2020;
- 3) Embase Ovid (1980 to 29 January, 2020);
- 4) Web of Science (Search date: 18 January, 2020);
- 5) CINAHL (EBSCOhost) (Search date: January, 2020);
- 6) SPORTDiscus (1949 to 10 January, 2020.

2.2.2 Searching other resources

We screened individual journals and conference proceedings (via handsearching). We also examined reference lists of all pertinent reviews and studies for published RCTs. We conducted a grey literature search to identify studies not indexed in the databases listed above. We used the following sources.

- 1) OpenGrey (<u>www.opengrey.eu</u>).
- 2) The web, e.g. Google (<u>http://scholar.google.com</u>)

2.3 Data collection and analysis

2.3.1 Selection of studies

Two investigators (EKW and MW) independently screened titles and abstracts using an online systematic review toolkit, Covidence (<u>https://www.covidence.org/</u>). Relevant studies were selected based on search queries. The full-text articles of studies identified as being potentially eligible for the review were retrieved to further determine their

inclusion. We resolved all disagreements by discussion between the review authors to reach a consensus. Authors applied the following inclusion criteria to determine if the full paper was needed for further scrutiny.

- 1) The study was a RCT or a cluster randomised trial and aimed to examine the effectiveness of a health provider based PA counselling/exercise prescription to increase PA levels
- 2) The study allocated participants to the intervention or control group/placebo group using a method of randomisation
- 3) The study included adults aged 18 years and older and recruited community dwelling adults that were free of chronic disease or pre-existing medical conditions that would limit participation in PA;
- 4) The study had analysed the results by intention to treat analysis or used per protocol analysis but with an attrition of less than a 20% (<u>Richards, Hillsdon, Thorogood & Foster, 2013</u>).

2.3.2 Data extraction and management

Two review authors (EKW and MW) independently carried out data extraction for each included record using a pro forma specifically designed for the purpose. We resolved differences in data extraction by discussion. The following characteristics from each study that met the inclusion criteria were extracted:

- 1) Description of study participants (e.g., sample size, demographic characteristics, country where study was performed);
- 2) Eligibility criteria for enrolment
- 3) Details about the intervention (e.g., length of intervention and follow-up, individual or group modality, behavioral techniques);
- 4) Assessment of risk of bias (e.g., study design, generation of allocation sequence, allocation concealment, blinding, loss to follow-up, inclusion of all randomised participants, incomplete outcome data addressed, and sample size calculation);
- 5) Outcome measures

We performed all statistical analyses using Review Manager 5.3 (<u>Review Manager</u> 5.3) software.



Figure 1: Study flow diagram

2.3.3 Assessment of risk of bias in included studies

Two review authors (MOO and EKW) assessed the risk of bias of included studies independently using Cochrane's 'Risk of bias' tool (Higgins, 2011). We assessed each study for risk of bias in each of the following domains: sequence generation, allocation concealment, blinding of participants and personnel (performance bias), blinding of outcome assessment (detection bias), incomplete outcome data and selective outcome reporting. We assessed the risk of bias associated with (a) blinding and (b) completeness of outcomes separately for self-reported outcomes and objective outcomes. Where there was disagreement between the review authors in risk of bias assessment, a third author (MW) was asked to independently appraise the study and discrepancies were resolved by consensus between all three authors. Risk was categorized as either "Low risk", "High risk", or "Unclear", and was listed in risk of bias tables broken down by trial. Studies at high risk of bias do appear in Data and analyses, but we suggest that readers use these data only to make decisions as to whether they would like to evaluate the intervention themselves in a more rigorous way. We do not believe the data support judgements about effect.

2.3.4 Measures of treatment effect

Statistical analysis was conducted according to Cochrane guidelines (<u>Higgins 2011</u>). We plotted the results of each trial as point estimates, using means and standard deviations (SDs) for the continuous outcomes. Since studies reported different outcome measures but measured the same concept, we calculated the standardised mean difference (SMD) with 95% confidence interval (CI) using a random-effects model. This strategy accounts for any potential heterogeneity that may occur following unique intervention approaches developed in various study settings.

2.3.5 Unit of analysis issues

We conducted statistical analysis using <u>Review Manager 5.3</u> software and visualised the results using forest plots. When a study had more than one study arm relevant to this review, we examined the overall effects of the intervention versus control by combining the data from the related study arms and calculated the standardized mean difference using recommended approaches (<u>Higgins, 2011</u>).We also calculated individual study effects and then the pooled effect sizes with 95% CIs. As in the 2013 update, we used Cohen's three levels to guide our classification of the estimates of effect as small (< 0.5), medium (0.5 to < 0.8) or large (V 0.8; <u>Cohen, 1988</u>). In studies where data from all repeated follow up assessments after baseline were available, we combined results from all follow up assessments to get an average change in outcomes for analysis.

The authors corrected the sample size in the cluster randomised trials included according to the Cochrane handbook (Higgins, 2011). A common approach is to use external estimates obtained from similar studies. According to the Cochrane handbook, for continuous data only the sample size need be reduced; means and standard deviations should remain unchanged. The effective sample size of a single intervention group in a cluster-randomised trial is its original sample size divided by a quantity called the 'design effect'. The design effect was calculated using the following formulae 1 + (M - 1) ICC. For this study we used ICC from a study by <u>Grandes et al., 2009</u> to compute the design effect.

2.3.6 Dealing with missing data

We applied the 'Risk of bias' criteria to exclude studies with a high risk of missing data, as they pose serious threats to validity (<u>Higgins, 2011</u>).Where appropriate, we contacted study authors for further information. If this was not possible, we reported the number of participants lost to follow-up. For studies that reported continuous data but did not report standard deviations, we calculated these values from other available data such as standard errors, or imputed them using the methods suggested in <u>Higgins 2011</u>.

2.3.7 Assessment of heterogeneity

First, we assessed whether studies were sufficiently homogeneous to be included in one comparison. We based this judgment on the similarity of the type of interventions, what the control condition was and the outcome. Statistical heterogeneity between results of

different studies was examined by χ^2 tests. A *P* value for a χ^2 test of less than 0.10 indicated heterogeneity. An alternative approach to quantify the effect of heterogeneity is assessing the inconsistency among the results of studies with 95% uncertainty intervals. A value of 0% indicates no observed heterogeneity and a value greater than 50% indicates the presence of substantial heterogeneity. We noted potential sources of differences between studies where analyses had high heterogeneity

2.3.8 Assessment of reporting biases

To reduce possible publication bias, we employed strategies to search for and identify relevant unpublished studies for inclusion. These strategies included searching the grey literature and prospective trial registration databases to overcome time-lag bias.

2.3.9 Data synthesis

We carried out statistical analysis using the Review Manager software (Review Manager 5.3). Had it been reasonable to assume that studies were estimating the same underlying treatment effect, i.e. where studies were examining the same intervention, and the studies' populations and methods were judged to be sufficiently similar, we would have used fixed-effect meta-analysis for combining data. Since there was clinical heterogeneity sufficient to expect that the underlying treatment effects differed between studies, we used random-effects meta-analysis to produce an overall summary when an average treatment effect across studies was considered clinically meaningful. Where the average treatment effect analyses, the results were presented as the average treatment effect with 95% confidence intervals

2.3.10 Subgroup analysis and investigation of heterogeneity

Although we identified heterogeneity with type of health provider (general practitioners and nurses), time of follow up, different interventions (face to face, telephone, Internet based) and with PA supervised verses unsupervised, we were unable to investigate subgroup analyses due to insufficient data.

2.3.11 Sensitivity analysis

We conducted sensitivity analyses to explore the effect of study quality, assessed by concealment of allocation, high attrition rates, or both, by excluding studies with a high or unclear risk of bias from the analyses in order to assess whether this made any difference to the overall result.

3. Results

3.1 Description of studies

The study selection process is summarized in <u>Figure 1</u> according to PRISMA guidelines. See <u>Characteristics of included studies</u>; <u>Characteristics of excluded studies</u>; <u>Characteristics of ongoing studies</u>

3.1.1 Results of the search

The database search identified 22, 210 studies. 12, 530 studies remained after duplicates were removed. We excluded 12,447 articles following a review of titles and abstracts and retrieved and assessed the full text of 55 articles. 24 randomised trials met the inclusion criteria, and we included them in this update. There were two ongoing studies which will be incorporated into the review at the next update. All searches were completed in January 2020.

3.1.2 Included studies

There were 15,269 apparently healthy adults who participated in the 24 included studies. All studies recruited both genders. The stated age range of participants was from 18 to 80 years, and six of the studies specifically targeted adults aged 65 years or older (<u>Goldstein et al., 1999</u>; <u>Kerse et al., 2005</u>; <u>Kolt et al., 2007</u>; <u>Mutrie et al., 2012</u>; <u>Petrella et al., 2003</u>; <u>Pfeiffer et al., 2001</u>)

3.1.3 Study design

Of the 24 included studies, 17 were randomised controlled trials (RCTs) (<u>Aittasalo</u>, <u>Miilunpalo</u>, <u>Harjula & Pasanen</u>, 2006; <u>Atay</u>, <u>Torman & Yaman</u>, 2014; <u>Fortier et al.</u>, 2011; <u>Green et al.</u>, 2002; <u>Hillsdon et al.</u>, 2002; <u>Kolt et al.</u>, 2007; <u>Lawton et al.</u>, 2008; <u>Lewis & Lynch</u>, 1993; <u>Little et al.</u>, 2004; <u>Mutrie et al.</u>, 2012; <u>Norris et al.</u>, 2000; <u>Pears et al.</u>, 2016; <u>Petrella et al.</u>, 2003; <u>Pfeiffer et al.</u>, 2001; <u>Reid & Morgan</u>, 1979; <u>Swinburn et al.</u>, 1998; <u>Sørensen et at.</u>, 2008; <u>Tylor & Fox</u>, 2005), including 7 cluster-RCTs (<u>Elley et al.</u>, 2003; <u>Goldstein et al.</u>, 1999; <u>Grandes et al.</u>, 2009; <u>Grandes et al.</u>, 2011; <u>Kerse et al.</u>, 2005; <u>Petrella</u>, <u>Lattanzio</u>, <u>Shapiro & Overend</u>, 2010)

3.1.4 Location and Setting

One study was conducted in Finland (<u>Aittasalo, Miilunpalo, Harjula & Pasanen, 2006</u>), one study in Turkey (<u>Atay, Torman & Yaman, 2014</u>), five studies in New Zealand (<u>Elley et al., 2003</u>; <u>Kerse et al., 2005</u>; <u>Kolt et al., 2007</u>; <u>Lawton et al., 2008</u>; <u>Swinburn et al., 1998</u>), two studies in Spain (<u>Grandes et al., 2009</u>; <u>Grandes et al., 2011</u>), four studies in the United States (<u>Goldstein et al., 1999</u>; <u>Green et al., 2002</u>; <u>Lewis & Lynch, 1993</u>; <u>Pfeiffer et al., 2001</u>), three studies in the United kingdom (<u>Hillsdon et al 2002</u>; <u>Little et al 2004</u>; <u>Mutrie et al., 2012</u>; <u>Pears et al., 2016</u>; <u>Tylor & Fox, 2005</u>), four studies in Canada (<u>Fortier et al 2011</u>; <u>Petrella, Lattanzio, Shapiro & Overend, 2010</u>; <u>Petrella et al., 2003</u>; <u>Reid & Morgan, 1979</u>),

one study in the Pacific north (<u>Norris et al 2000</u>) and one study in Denmark (<u>Sørensen et at 2008</u>).

3.1.5 Types of intervention

The interventions in the study were all based in a primary care setting and were administered by either physician, general practitioners, doctors or nurses. The main modes of delivery of physical activity counselling were face to face, telephone, written or combined. Of the 24 included studies, 21 were face to face delivery (<u>Aittasalo, Miilunpalo, Harjula & Pasanen, 2006; Atay, Torman & Yaman, 2014; Elley et al., 2003; Fortier et al., 2011; Goldstein et al., 1999; Grandes et al., 2011; Hillsdon et al., 2002; Kerse et al., 2005; Lawton et al., 2008; Lewis & Lynch, 1993; Little et al., 2004; Mutrie et al., 2012; Norris et al., 2000; Pears et al., 2016; Petrella, Lattanzio, Shapiro & Overend, 2010; Petrella et al., 2003; Pfeiffer et al., 2001; Reid & Morgan, 1979; Sørensen et at., 2008; Tylor & Fox, 2005), five were either telephone delivered or combined with face to face (Green et al 2002; Hillsdon et al., 2002; Kolt et al., 2007; Lawton et al., 2008; Swinburn et al., 1998), 5 were either written only or combined with face to face delivery (Elley et al., 2003; Grandes et al., 2009; Lewis & Lynch, 1993; Little et al., 2004; Reid & Morgan, 1979).</u>

3.1.6 Follow up periods

The follow up periods ranged from one month to 25 months. One study used a one month follow up (Little et al., 2004), five studies used a six months follow up period (Aittasalo, Miilunpalo, Harjula & Pasanen, 2006; Atay, Torman & Yaman, 2014; Grandes et al., 2009; Norris et al 2000; Reid & Morgan, 1979), five studies used a 12 months follow up period (Hillsdon et al., 2002; Kerse et al., 2005; Kolt et al., 2007; Petrella, Lattanzio, Shapiro & Overend, 2010; Petrella et al., 2003), and 3 studies used a 24 months follow up period (Grandes et al., 2011; Green et al., 2002; Lawton et al., 2008). Other studies had variables follow up periods (Elley et al., 2003; Fortier et al., 2011; Goldstein et al., 1999; Lewis & Lynch, 1993; Mutrie et al., 2012; Pears et al., 2016; Pfeiffer et al., 2001; Swinburn et al., 1998; Sørensen et al., 2008; Tylor & Fox, 2005)

3.1.7 Outcomes

The primary out comes in the studies were: Total energy expenditure (Kcal/week or Kcal/day) reported in 8 studies (<u>Atay, Torman & Yaman, 2014</u>; <u>Elley et al., 2003</u>; <u>Hillsdon et al., 2002</u>; <u>Kerse et al., 2005</u>; <u>Norris et al., 2000</u>; <u>Pears et al., 2016</u>; <u>Petrella, Lattanzio, Shapiro & Overend, 2010</u>; <u>Tylor & Fox, 2005</u>), Self reported physical activity (minutes/week or hours/week) reported in 17 studies (<u>Aittasalo, Miilunpalo, Harjula & Pasanen, 2006</u>; <u>Elley et al 2003</u>; <u>Goldstein et al., 1999</u>; <u>Grandes et al., 2009</u>; <u>Grandes et al., 2008</u>; <u>Lewis & Lynch, 1993</u>; <u>Little et al., 2004</u>; <u>Norris et al., 2000</u>; <u>Pears et al., 2016</u>; <u>Pfeiffer et al., 2001</u>; <u>Swinburn et al., 1998</u>; <u>Sørensen et at 2008</u>; <u>Tylor & Fox, 2005</u>). Aerobic fitness VO2 Max (ml/kg/min or ml/min) reported in six studies (<u>Fortier et al., 2011</u>; <u>Grandes et al., 2009</u>; <u>Max (ml/kg/min or ml/min) reported in six studies (Fortier et al., 2011</u>; <u>Petrella et al., 2009</u>; <u>Petrella et al., 2003</u>; <u>Petrella et al., 2003</u>

Reid & Morgan, 1979), Systolic blood pressure(mmHg) reported in seven studies (Atay, Torman & Yaman, 2014; Elley et al., 2003; Hillsdon et al., 2002; Kerse et al., 2005; Lawton et al., 2008; Petrella, Lattanzio, Shapiro & Overend, 2010; Petrella et al., 2003) and Body Mass Index (BMI) reported in seven studies (Atay, Torman & Yaman, 2014; Fortier et al., 2011; Hillsdon et al., 2002; Petrella, Lattanzio, Shapiro & Overend, 2010; Petrella et al., 2003; Sørensen et at., 2008; Tylor & Fox, 2005). Other reported outcomes include Steps/walking, Body circumferences (waist, hip or limb), Total cholesterol and Blood Glucose.

3.1.8 Excluded studies

After assessment, the following 31 studies were excluded: 3 did not have a control group (Activity, 2001; Armit et al., 2009; Hardcastle, 2012); 2 did not use a randomised control trial study/cluster randomised control trial design (Bucholz & Purath, 2007; Burn, Camaione & Chartterton, 2000); 3 reported the outcomes as odds ratios and could not be included in the meta analysis (Gao et al, 2016; Harland et al, 1999; Harrison, Roberts & Elton, 2004); 15 were not based in a primary care setup and intervention not health provider based (Bennet et al, 2008; Calfas et al, 1996; Cunningham et al, 1987; Duru et at., 2010; Kriska et al., 1986; Marcus et al., 2007; Pekmezi et al., 2016; Schröder et al,. 2018; Skår et al., 2011; Spittaels, Bourdeaudhuij & Vandelanotte, 2007; Steele, Mummery & Dwyer, 2009; Stralen et al., 2009; Wanner et al., 2009; William et al., 2006); 4 had inadequate data and attempts to reach the authors were futile (Bull & Jamrozik, 1999; Dubbert et al., 2002; Fortier et al., 2006; Jimmy & Martin, 2005); 5 reported different outcomes and/or used different interventions (Kastarinen et al., 2002; Long et al., 1996; Nymberg et al., 2018; Pinto et al., 1998; Rome, Ekdahl & Gard, 2009). See Characteristics of excluded studies.

3.2 Risk of bias in included studies

We prepared an assessment of risk of bias for each trial and illustrated final judgements for the ten criteria in Figure 2 and Figure 3. All studies had some methodological shortcomings, in most instances related to unclear risk of bias for different criteria.



Figure 2: Risk of bias graph: review authors' judgements about

3.2.1 Allocation (selection bias)

Allocation bias was generally low in the review with 92% (22) of the studies having low risk. Only one study had a high risk of allocation bias. Most studies stated that participants or practices (in case of cluster randomisation) were assigned randomly, according to a computerised randomisation scheme. For one study, the risk of bias for random sequence generation was unclear owing to poor reporting.



Figure 3: Risk of bias summary: review authors' judgements about each risk of bias item for each included study

3.2.2 Blinding (performance bias and detection bias)

We judged the risk of performance bias as unclear because no information was available. Of the 24 studies included, 33% (8) studies had low risk, while 54% (13) were classified as having unclear risk because they did not provide sufficient information on blinding of outcome assessment. Only three studies had high risk of performance bias. We expect that patients and personnel were not blinded in these studies because the care provider constitutes the intervention. Whether this lack of blinding influences outcomes is unclear.

3.2.3 Incomplete outcome data (attrition bias)

In half of the studies, 80% or more of the initial participants completed the study. Risk of bias due to incomplete outcome data was unclear in six of the studies because of limited reporting about follow-up. Four studies were classified as having high risk of attrition bias due to high follow up losses and no explanation at all.

3.2.4 Selective reporting (reporting bias)

A protocol was available for each study, and 63% (15) of the studies reported predefined outcome measures. Absence of study protocols to confirm reporting of all intended outcomes led to the unclear judgement in seven of the studies.



Note: Funnel plot of comparison: 1 Physical activity counselling in primary care setting verses Usual care or placebo, outcome: 1.1 Self reports physical activity (minutes/week or hours/week)



Note: Funnel plot of comparison: 1 Physical activity counselling in primary care setting verses Usual care or placebo, outcome: 1.2 Total energy expenditure (Kcal/week or Kcal/day).



Note: Funnel plot of comparison: 1 Physical activity counselling in primary care setting verses Usual care or placebo, outcome: 1.5 Systolic blood pressure(mmHg). Effects of interventions

3.2.5 Publication bias

Meta-analysis may be vulnerable to publication bias if studies with less favourable results are excluded. A useful test for publication bias is based on the funnel plot, which compares intervention effects estimated from individual studies against a measure of study size. In the absence of bias, the plot resembles a symmetrical inverted funnel (<u>Sterne, 2001</u>).

In the current review, there was no clear evidence of funnel plot asymmetry for the main significant outcomes total energy expenditure (Kcal/week or Kcal/day) and systolic blood pressure (mmHg) (Figure 7; Figure 11), however there was some signs of publication bias on the outcome self reported physical activity (minutes/week or hours/week), this can be attributed to limited studies with small sample size. See funnel plot (Figure 5).

A total of 23 trials investigated patient outcomes (<u>Aittasalo, Miilunpalo, Harjula &</u> Pasanen, 2006; <u>Atay, Torman & Yaman, 2014</u>; <u>Elley et al., 2003</u>; <u>Fortier et al., 2011</u>; <u>Goldstein et al., 1999</u>; <u>Grandes et al., 2009</u>; <u>Grandes et al., 2011</u>; <u>Green et al., 2002</u>; <u>Hillsdon et al., 2002</u>; <u>Kerse et al., 2005</u>; <u>Kolt et al., 2007</u>; <u>Lawton et al., 2008</u>; <u>Lewis & Lynch, 1993</u>; <u>Little et al., 2004</u>; <u>Norris et al., 2000</u>; <u>Pears et al., 2016</u>; <u>Petrella, Lattanzio, Shapiro &</u> <u>Overend, 2010</u>; <u>Petrella et al., 2003</u>; <u>Pfeiffer et al., 2001</u>; <u>Reid & Morgan, 1979</u>; <u>Swinburn et al., 1998</u>; <u>Sørensen et al., 2008</u>; <u>Tylor & Fox, 2005</u>)

We grouped patient outcomes into the following categories: self reported physical activity (minutes/week or hours/week), Total energy expenditure (Kcal/wee or Kcal/day), aerobic fitness VO2 max (ml/kg/min or ml/min), body mass index (Kg/m²), systolic blood pressure (mmHg), total cholesterol (mmol/l) and waist circumference.

Patient or populati	Patient or population: primary care: A systematic review and meta-analysis												
Setting: primary ca	Setting: primary care setting												
Intervention: Phys	Intervention: Physical activity counselling in primary care setting												
Comparison: Usual care or placebo													
Outcomes	Anticipated absolute		Relative	Nº of	Certainty of	Comments							
	effects* (95% CI)		effect	participants	the evidence								
			(95% CI)	(studies)	(GRADE)								
	Risk with placebo	Risk with											
		Physical											
		activity											
		counselling											
		in primary											
		care setting											
		versus											
		Usual care											
		or placebo											
Self reports	The mean self	SMD 0.11	-	13211	$\Theta \Theta \Theta \Theta$								
physical activity	reports physical	higher		(16 RCTs)	LOW 12345								
(minutes/week or	activity	(0.04 higher											
hours/week)	(minutes/week or	to 0.17											
	hours/week) was 0	higher)											
Total energy	The mean total	SMD 0.2	-	3376	$\oplus \oplus \oplus \oplus$								
expenditure	energy expenditure	higher		(8 RCTs)	HIGH ¹³⁴⁶								
(Kcal/week or	(Kcal/week or	(0.13 higher											
Kcal/day)	Kcal/day) was 0	to 0.27											
		higher)											
Aerobic fitness	The mean aerobic	SMD 0.06	-	8822	$\oplus \oplus \ominus \ominus$								
VO2 Max	fitness VO2 Max	higher		(6 RCTs)	LOW 3789								
(ml/kg/min or	(ml/kg/min or	(0.01 lower											
ml/min)	ml/min) was 0	to 0.12											
		higher)											

Tables 1: Summary of findings

Body Mass Index	The mean body	SMD 0.04	-	2371	$\oplus \Theta \Theta \Theta$	
(BMI)	Mass Index (BMI)	lower		(7 RCTs)	VERY LOW ³	
	was 0	(0.15 lower			9 10 11	
		to 0.07				
		higher)				
Systolic blood	The mean systolic	SMD 0.1	-	2856	$\oplus \oplus \oplus \ominus$	
pressure(mmHg)	blood	lower		(7 RCTs)	MODERATE	
1	pressure(mmHg)	(0.18 lower			3 4 12 13	
	was 0	to 0.03				
		lower)				
Total cholesterol	The mean total	SMD 0.27	-	2459	$\Theta \Theta \Theta \Theta$	
	cholesterol was 0	lower		(4 RCTs)	VERY LOW 1	
		(0.72 lower			3 11 14 15 16	
		to 0.18				
		higher)				
Waist	The mean waist	SMD 0.05	-	1469	$\Theta \Theta \Theta \Theta$	
circumference	circumference was 0	lower		(4 RCTs)	VERY LOW ³	
		(0.15 lower			9 12 16 17	
		to 0.06				
		higher)				

*The risk in the intervention group (and its 95% confidence interval) is based on the assumed risk in the comparison group and the **relative effect** of the intervention (and its 95% CI). CI: Confidence interval; **RR**: Risk ratio; **OR**: Odds ratio;

Grade Working Group grades of evidence

High certainty: We are very confident that the true effect lies close to that of the estimate of the effect **Moderate certainty:** We are moderately confident in the effect estimate: the true effect is likely to be close to the estimate of the effect, but there is a possibility that it is substantially different

Low certainty: Our confidence in the effect estimate is limited: the true effect may be substantially different from the estimate of the effect

Very low certainty: We have very little confidence in the effect estimate: the true effect is likely to be substantially different from the estimate of effect

¹ The studies with the highest weights only had unclear performance bias (single-blind). However, some studies noted that it was impossible to blind the personnel in this kind of intervention

 2 The variation of effect sizes in the study was too big (0.02-0.46). The I2 was moderate at 50% and the P-value of heterogeneity was statistically significant (p=0.01)

³ A few variations in the intervention between studies. Some studies have a combined intervention of written and face to face counselling, telephone, and face to face counselling while others only focus on face to face,

⁴ The confidence interval of the overall effect was not wide and the sample size was above the recommended of 400 and the pooled effect did not include a null effect

⁵ The funnel plot was asymmetric indicating publication bias

⁶ There was a good overlap of confidence intervals amongst studies, the I2 was low at 2% and the P-value of heterogeneity was not statistically significant (P=0.42)

⁷ Only one study has signs of performance, detection, and attrition bias

⁸ There was a good overlap of confidence intervals amongst studies, the I2 was moderate at 35% and the P-value of heterogeneity was not statistically significant (P=0.17)

⁹ The confidence interval of the overall effect included the null effect and it was unclear if there was benefit or harm

¹⁰ There was a high performance, detection, and attrition risk of bias in the studies that had the highest weight

¹¹ There were huge variations in the size and direction of the effects in the study (0.01 to-0.35). There was also a poor overlap of confidence intervals

¹² The studies with high weight had an unclear performance, detection and attrition risk of bias

¹³ There was a good overlap of confidence intervals amongst studies, all studies had consistency in direction of the effects, the I2 was low at 0% and the p-value of heterogeneity was not statistically significant (p=0.54)

¹⁴ There was little to no overlap amongst confidence intervals, the I2 percentage was 96% indicating heterogeneity amongst studies

¹⁵ There were huge variations in the size and direction of the effects in the study (0.09 to-0.75). There was also a poor overlap of confidence intervals

¹⁶ The studies were below 10 hence we could not generate a funnel plot, however, some studies had signs of selective analysis and poor methodology

¹⁷ There was a good overlap of confidence intervals amongst studies, all studies had consistency in direction of the effects, the I2 was low at 0% and the p-value of heterogeneity was not statistically significant (p=0.42)

Note: Physical activity counselling in primary care setting versus usual care or placebo compared to placebo in primary care: a systematic review and meta-analysis

3.2.6 Self reported physical activity (minutes/week or hours/week)

Sixteen trials evaluated self reported physical activity in minutes/week (<u>Aittasalo</u>, <u>Miilunpalo</u>, <u>Harjula & Pasanen</u>, 2006; <u>Elley et al.</u>, 2003; <u>Fortier et al.</u>, 2011; <u>Goldstein et al.</u>, 1999; <u>Grandes et al.</u>, 2009; <u>Grandes et al.</u>, 2011; <u>Green et al.</u>, 2002; <u>Kerse et al.</u>, 2005; <u>Kolt et al.</u>, 2007; <u>Lawton et al.</u>, 2008; <u>Little et al.</u>, 2004; <u>Norris et al.</u>, 2000; <u>Pears et al.</u>, 2016; <u>Pfeiffer et al.</u>, 2001; <u>Swinburn et al.</u>, 1998; <u>Sørensen et al.</u>, 2008). Meta-analysis of data from these trials suggests that health provider-led physical activity counselling interventions in primary care may lead to increased physical activity among patients. The pooled effect was positive and high favouring the physical activity counselling interventions over usual care/placebo (random effects analysis P=0.0008). Data show little evidence of statistical heterogeneity. Excluding from the meta-analysis a trial by (<u>Lewis & Lynch</u>, 1993) greatly changed the result. Results differed considerably in the sensitivity analyses (SMD 0.11, 95% CI 0.04 to 0.17; participants = 13211; studies = 16; I² = 50% Figure <u>4</u>). A SMD of 0.11 represents a small difference between groups. The evidence is of low certainty (<u>Summary of findings table 1</u>) owing to a small confidence interval and moderate clinical heterogeneity, see, <u>Analysis 1.1</u>.

					Figure 4	: Ana	lysis 1.	1		
	PA	counselling		Usual	care or place	bo		Std. Mean Difference	Std. Mean Difference	Risk of Bias
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% CI	ABCDEF
Aittasalo, Millunpalo, Harjula & Pasanen, 2006	548	633.2061	130	488	755.1338	73	3.6%	0.09 [-0.20, 0.37]		2 3 2 3 3 3
Elley et al 2003	54.6	142.641	451	16.8	113.5413	427	9.5%	0.29 [0.16, 0.43]		
Fortier et al 2011	25.7	18.5	61	22.4	19.7	59	2.5%	0.17 [-0.19, 0.53]		
Goldstein et al 1999	112.8	72.7792	158	111.3	68.8737	154	5.3%	0.02 [-0.20, 0.24]		
Grandes et al 2009	82.58	547.3845	2248	65.14	527.4343	2069	14.6%	0.03 [-0.03, 0.09]		
Grandes et al 2011	148.82	743.7275	1906	139.97	721.21	1785	14.3%	0.01 [-0.05, 0.08]		
Green et al 2002	5.37	1.5952	128	4.98	1.5952	128	4.6%	0.24 [-0.00, 0.49]		2 3 3 3 2 2
Kerse et al 2005	1.97	2.0945	130	1.3	2.0945	140	4.7%	0.32 [0.08, 0.56]		
Kolt et al 2007	244	365.7	93	117.3	138.8	93	3.6%	0.46 [0.16, 0.75]		
Lawton et al 2008	105	1,187.3596	544	90	1,188.4553	545	10.4%	0.01 [-0.11, 0.13]		
Lewis & Lynch, 1993	108.67	103.5044	66	-23.7	103.5044	97		Not estimable		
Little et al 2004	10.47	24.2718	182	7	47.7265	190	6.0%	0.09 [-0.11, 0.29]		
Norris et al 2000	201.8	161.1533	450	186.9	161.1533	362	9.1%	0.09 [-0.05, 0.23]		
Pears et al 2016	75.8	25.8619	52	73.1	32.4294	111	2.9%	0.09 [-0.24, 0.42]	2	
Pfeiffer et al 2001	320	714.6669	25	223	504.425	24	1.1%	0.15 [-0.41, 0.71]	-	
Swinburn et al 1998	156	231.8077	218	156	342.4849	238	6.8%	0.00 [-0.18, 0.18]		
Sørensen et at 2008	28.8024	130.2997	19	18.8024	130.2997	18	0.9%	0.08 [-0.57, 0.72]		→ ●●?●●?
Total (95% CI)			6795			6416	100.0%	0.11 [0.04, 0.17]	•	
Heterogeneity: Tau ² = 0.01; Chi ² = 30.24, df = 15 Test for overall effect: Z = 3.35 (P = 0.0008)	(P = 0.01);	P= 50%							-0.5 -0.25 0 0.25 0 Favours usual care Favours PA cours	1.5 selling
Risk of bias legend										
(A) Random sequence generation (selection bia	as)									
(B) Allocation concealment (selection bias)										
(C) Blinding of participants and personnel (perfo	ormance bia	IS)								
(D) Blinding of outcome assessment (detection	bias)									
(E) Incomplete outcome data (attrition bias)										
(F) Selective reporting (reporting bias)										

Notes: Forest plot of comparison: 1 Physical activity counselling in primary care setting verses Usual care or placebo, outcome: 1.1 Self reports physical activity (minutes/week or hours/week).

3.2.7 Total energy expenditure (Kcal/week or Kcal/day)

We found high-quality evidence, based on eight studies (<u>Atay, Torman & Yaman, 2014</u>; <u>Elley et al., 2003</u>; <u>Hillsdon et al., 2002</u>; <u>Kerse et al., 2005</u>; <u>Norris et al., 2000</u>; <u>Pears et al., 2016</u>; <u>Petrella, Lattanzio, Shapiro & Overend, 2010</u>; <u>Tylor & Fox, 2005</u>), that the use health provider-led physical activity counselling interventions in primary care may lead to increased total energy expenditure among patients when compared to usual care alone (SMD 0.20, 95% CI 0.13 to 0.27; participants = 3376; studies = 8; I² = 2%; overall effect P=0.00001) Figure 6. A SMD of 0.2 represents a small difference between groups and data show no evidence of statistical heterogeneity. See <u>Analysis 1.2</u>:

Figure 6: Analysis 1.2											
	PA counselling Usual care or placebo			ebo	1	Std. Mean Difference	Std. Mean Difference	Risk of Bias			
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl	ABCDEF	
Atay, Torman & Yaman, 2014	34.52	1.24	69	33.95	1.1	110	5.1%	0.49 [0.19, 0.80]			
Elley et al 2003	9.76	42.252	451	0.37	39.5292	427	26.1%	0.23 [0.10, 0.36]			
Hillsdon et al 2002	148	273.7583	302	98	197.2638	285	17.7%	0.21 [0.05, 0.37]		••••	
Kerse et al 2005	220.96	24.6311	130	218	24.6311	140	8.3%	0.12 [-0.12, 0.36]			
Norris et al 2000	2,108	155.0828	362	2,047	637.1749	450	24.1%	0.13 [-0.01, 0.26]		😌 😌 ? ? 🗨 🚭	
Pears et al 2016	33	18.0352	59	32.2	22.1291	120	4.9%	0.04 [-0.27, 0.35]			
Petrella, Lattanzio, Shapiro & Overend, 2010	0.67	1.5804	169	0.25	1.1528	160	10.0%	0.30 [0.08, 0.52]		•••	
Tylor & Fox, 2005	33.9	1.9698	97	33.4	2.6833	45	3.8%	0.22 [-0.13, 0.58]		• • • • • • • •	
Total (05% CD			4620			4727	100.0%	0 20 10 42 0 271			
		17 0.07	1039			11.51	100.0%	0.20 [0.13, 0.27]	- C - C - C - C - C		
Heterogeneity: Taur = 0.00, Chir = 7.12, df = 7 (P = 0.42);	r= 2%							-1 -0.5 0 0.5 1		
Test for overall effect: $Z = 5.74$ (P < 0.00001)									Favours usual care Favours PA counsel	ling	
Disk of bios lossed											
(A) Denders exercise (aslastics /	ingl										
(A) Random sequence generation (selection b	ias)										
(B) Allocation concealment (selection bias)		kine)									
(C) Binding of participants and personnel (per	iormance	blas)									
(D) Binding of outcome assessment (detection	i bias)										
(E) incomplete outcome data (attrition bias)											
(r) Selective reporting (reporting bias)											

Notes: Forest plot of comparison: 1 Physical activity counselling in primary care setting verses Usual care or placebo, outcome: 1.2 Total energy expenditure (Kcal/week or Kcal/day).

3.2.8 Aerobic fitness VO2 Max (ml/kg/min or ml/min)

Sixt trials (8822 participants) evaluated aerobic fitness as VO2 max (Fortier et al., 2011; Grandes et al., 2009; Grandes et al., 2011; Petrella et al., 2003; Petrella, Lattanzio, Shapiro & Overend, 2010; Reid & Morgan, 1979). Meta-analysis of data from these trials suggests that health provider-led physical activity counselling interventions in primary care did not lead to increased aerobic fitness among patients(SMD 0.06, 95% CI -0.01 to 0.12; I² = 35%) Figure 8. Data show moderate evidence of statistical heterogeneity. The pooled effect

was not significant (random effects analysis P=0.11). The evidence is of low certainty (<u>Summary of findings, Table 1</u>) owing to a large confidence interval and moderate clinical heterogeneity, see, <u>Analysis 1.3</u>:

	PAc	ounselling	g	Usual care or placebo			Std. Mean Difference	Std. Mean Difference	Risk of Bias	
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% Cl	IV, Random, 95% Cl	ABCDEF
Fortier et al 2011	28.246	4.208	61	27.2	4.208	59	3.4%	0.25 [-0.11, 0.61]		
Grandes et al 2009	1.18	8.2204	2248	1.09	8.1179	2069	40.1%	0.01 [-0.05, 0.07]	+	$\bullet \bullet ? \bullet \bullet \bullet$
Grandes et al 2011	1.06	12.2433	1906	0.86	12.0632	1785	38.2%	0.02 [-0.05, 0.08]	+	$\bullet \bullet ? \bullet \bullet \bullet$
Petrella et al 2003	24.9	10.301	131	22.8	13.6345	110	6.4%	0.18 [-0.08, 0.43]		
Petrella, Lattanzio, Shapiro & Overend, 2010	3.02	4.0827	169	2.21	6.0203	160	8.5%	0.16 [-0.06, 0.37]	+	••?
Reid & Morgan, 1979	12.44	7.6627	77	9.66	7.6627	47	3.3%	0.36 [-0.01, 0.73]	1 * · ·	•••
Total (95% Cl)45924230100.0%0.06 [-0.01, 0.12]Heterogeneity: Tau ² = 0.00; Chi ² = 7.70, df = 5 (P = 0.17); I ² = 35% $-0.5 - 0.25$)	
 (C) Billinding of participants and personnel (period) (D) Blinding of outcome assessment (detection (E) Incomplete outcome data (attrition bias) (F) Selective reporting (reporting bias) 	n bias)	uias)								

Figure 8: Analysis 1.3

Notes: Forest plot of comparison: 1 Physical activity counselling in primary care setting verses Usual care or placebo, outcome: 1.3 Aerobic fitness VO2 Max (ml/kg/min or ml/min).

3.2.9 Body Mass Index (BMI) (Kg/m²)

Seven trials (2371 participants) investigated changes in body mass index (<u>Atay, Torman & Yaman, 2014</u>; <u>Elley et al., 2003</u>; <u>Fortier et al., 2011</u>; <u>Hillsdon et al., 2002</u>; <u>Petrella et al., 2003</u>; <u>Petrella, Lattanzio, Shapiro & Overend, 2010</u>; <u>Sørensen et at., 2008</u>; <u>Tylor & Fox, 2005</u>). All the trials provided sufficient data for a meta-analysis. Data show that there may be little or no difference in body mass index (SMD - 0.04, 95% CI -0.15 to 0.07) Figure 9</u>. For body mass index, the evidence is of very low certainty (<u>Summary of findings, Table 1</u>) owing to moderate inconsistency (I² = 34%; overall effect P=0.11; Random effects analysis) and imprecision (wide confidence interval). Results did not change considerably under <u>Sensitivity analysis</u>. See, <u>Analysis 1.4</u>:



Figure 9: Analysis 1.4

Notes: Forest plot of comparison: 1 Physical activity counselling in primary care setting verses Usual care or placebo, outcome: 1.4 Body Mass Index (BMI).

3.2.10 Systolic blood pressure (mmHg)

Eight trials (2856 participants) reporting systolic blood pressure at six showed a mean difference (MD) favouring health provider-led physical activity counselling interventions compared to usual care of -0.10 mmHg (95% CI -0.18 to -0.03; <u>Analysis 1.5</u>. The analysis was based on a fixed effects model. One study (<u>Lawton et al., 2008</u>) was excluded during sensitivity analysis. There was no heterogeneity amongst results of the remaining trials (<u>Atay, Torman & Yaman, 2014</u>; <u>Elley et al., 2003</u>; <u>Hillsdon et al., 2002</u>; <u>Kerse et al., 2005</u>; <u>Little et al., 2004</u>; <u>Petrella et al., 2003</u>; <u>Petrella, Lattanzio, Shapiro & Overend, 2010</u>) (I² = 0%, overall effect P = 0.006) <u>Figure 10</u>. The evidence is of moderate certainty owing to a small confidence interval and small clinical heterogeneity (<u>Summary of findings, Table 1</u>).

	PA c	ounselling	J	Usual o	are or plac	ebo	5	td. Mean Difference	Std. Mean Difference	Risk of Bias
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Fixed, 95% CI	IV, Fixed, 95% Cl	ABCDEF
Atay,Torman & Yaman, 2014	127.89	16.88	69	131.93	18.46	110	5.9%	-0.23 [-0.53, 0.08]		
Elley et al 2003	-2.58	15.5608	451	-1.21	14.2978	427	30.9%	-0.09 [-0.22, 0.04]		$\bullet \bullet \bullet \bullet \bullet \bullet \bullet$
Hillsdon et al 2002	-3.2	12.3633	302	-2.9	12.865	285	20.7%	-0.02 [-0.19, 0.14]		•••
Kerse et al 2005	142.45	23.7514	130	143	23.7514	140	9.5%	-0.02 [-0.26, 0.22]		
Lawton et al 2008	119.1	0.7	544	119.5	0.7	545		Not estimable		•••••
Little et al 2004	-5.29	18.8021	182	-3.86	39.411	190	13.1%	-0.05 [-0.25, 0.16]		••??••
Petrella et al 2003	127	45.7821	131	137	31.4643	110	8.4%	-0.25 [-0.50, 0.00]		
Petrella, Lattanzio, Shapiro & Overend, 2010	-4.13	14.8821	169	-0.38	16.652	160	11.5%	-0.24 [-0.45, -0.02]		•••?
Total (95% CI) Heterogeneity: Chi ² = 5.06, df = 6 (P = 0.54); I ² = Test for overall effect: Z = 2.74 (P = 0.006)	:0%		1434			1422	100.0%	-0.10 [-0.18, -0.03]	-0.5 -0.25 0 0.25 0.5 Favours PA counselling Favours usual care	-
Risk of bias legend (A) Random sequence generation (selection bias) (B) Allocation concealment (selection bias) (C) Blinding of participants and personnel (performance bias) (D) Blinding of outcome assessment (detection bias) (E) Incomplete outcome data (attrition bias) (F) Selective reporting (reporting bias)										

Figure 10: Analysis 1.5

Notes: Forest plot of comparison: 1 Physical activity counselling in primary care setting verses Usual care or placebo, outcome: 1.5 Systolic blood pressure(mmHg).

3.2.11 Total cholesterol (mmol/l)

Four trials (2459 participants) investigated total cholesterol (<u>Elley et al., 2003</u>; <u>Fortier et al., 2011</u>; <u>Lawton et al., 2008</u>; <u>Little et al., 2004</u>). The meta-analysis suggests that there may be little or no difference between health provider-led physical activity counselling interventions in primary care and usual care (SMD -0.27, 95% CI -0.72 to 0.18; very low-certainty evidence; <u>Analysis 1.6</u>). For total cholesterol, evidence is of very low certainty (<u>Summary of findings, Table 1</u>) owing to strong inconsistency (I² = 96%; overall effect P=0.24; random effects analysis <u>Figure 12</u>) and a wide confidence interval, suggesting that the extent to which the total cholesterol differs between health provider-led physical activity counselling interventions in primary care and usual care varied with the context of physical activity. Results did not change considerably in the <u>Sensitivity analysis</u>.

	PAc	ounsellin	ng	Usual c	Usual care or placebo St			Std. Mean Difference	Std. Mean Difference	Risk of Bias	
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Random, 95% CI	IV, Random, 95% Cl	ABCDEF	
Elley et al 2003	-0.019	0.6592	451	0.01	0.6308	427	26.0%	-0.04 [-0.18, 0.09]	+		
Fortier et al 2011	4.453	0.651	61	4.7	0.651	59	22.7%	-0.38 [-0.74, -0.02]			
Lawton et al 2008	-5.86	0.04	544	-5.83	0.04	545	26.1%	-0.75 [-0.87, -0.63]	•		
Little et al 2004	-0.05	0.8205	182	-0.34	4.1927	190	25.2%	0.09 [-0.11, 0.30]	*	$\bullet \bullet ? ? \bullet \bullet$	
Total (95% CI)			1238			1221	100.0%	-0.27 [-0.72, 0.18]	•		
Heterogeneity: Tau ² = 0.20; Chi ² = 79.34, df = 3 (P < 0.00001); l ² = 96%											
Test for overall effect: Z = 1.18 (P = 0.24) -4 -2 0 2 4 Favours PA counselling Favours usual care											
Risk of bias legend (A) Random sequence generation (selection bias) (B) Allocation concealment (selection bias) (C) Blinding of participants and personnel (performance bias) (D) Blinding of outcome assessment (detection bias)											
(E) Incomplete outcom (F) Selective reporting	me data (g (reportir	attrition b ng bias)	oias)								

Figure 12: Analysis 1.6

Notes: Forest plot of comparison: 1 Physical activity counselling in primary care setting verses Usual care or placebo, outcome: 1.6 Total cholesterol.

3.2.12 Waist circumference (cm)

Four trials (1469 participants) investigated waist circumference (<u>Atay, Torman & Yaman, 2014</u>; <u>Fortier et al., 2011</u>; <u>Lawton et al., 2008</u>; <u>Tylor & Fox, 2005</u>). All the trials provided sufficient data for a meta-analysis. The meta-analysis suggests that there may be little or no difference between health provider-led physical activity counselling interventions in primary care and usual care (SMD -0.05, 95% CI - 0.15 to 0.06; very low-certainty evidence; <u>Analysis 1.6</u>). For total cholesterol, evidence is of very low certainty (<u>Summary of findings, Table 1</u>) owing to a wide confidence interval (imprecision), suggesting that the extent to which the waist circumference differs between health provider-led physical activity counselling interventions in primary care and usual care varied with the context of physical activity. There was no heterogeneity amongst study results (I² = 0%; overall effect P=0.24; fixed effects analysis). Results did not change considerably in the <u>Sensitivity analysis</u>.

Figure 13: Analysis 1.8												
	PAc	ounselling	J	Usual care or placebo			1	Std. Mean Difference	Std. Mean Difference	Risk of Bias		
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	Weight	IV, Fixed, 95% Cl	IV, Fixed, 95% Cl	ABCDEF		
Atay, Torman & Yaman, 2014	96.75	15.02	69	98.48	11.61	110	11.6%	-0.13 [-0.43, 0.17]		•••••••••••••••••••••••••••••••••••••••		
Fortier et al 2011	100.427	3.9593	61	101.1	3.9593	59	8.2%	-0.17 [-0.53, 0.19]				
Lawton et al 2008	88.7	0.6	544	88.7	0.6	545	74.8%	0.00 [-0.12, 0.12]	- -	••••		
Tylor & Fox, 2005	90.2	15.5994	46	94.7	11.2406	35	5.4%	-0.32 [-0.76, 0.12]		😠 👁 ? ? 👁 ?		
Tylor & Fox, 2005 90.2 15.5994 46 94.7 11.2406 35 5.4% -0.32 [-0.76, 0.12] Total (95% Cl) 720 749 100.0% -0.05 [-0.15, 0.06] -0.5 -0.5 0.25 0 0.25 0.5 0.5 0.25 0.5												
 (D) Blinding of outcome asses (E) Incomplete outcome data (a (F) Selective reporting (reporting) 	 (C) Blinding of participants and personnel (performance bias) (D) Blinding of outcome assessment (detection bias) (E) Incomplete outcome data (attrition bias) (F) Selective reporting (reporting bias) 											

Notes: Forest plot of comparison: 1 Physical activity counselling in primary care setting verses Usual care or placebo, outcome: 1.8 Waist circumference.

4. Data and analyses

<u> </u>	0	1 /	1	
Outcome or Subgroup	Studies	Participants	Statistical Method	Effect
				Estimate
1.1 Self reports physical activity	16	13211	Std. Mean Difference	0.11
(minutes/week or hours/week)			(IV, Random, 95% CI)	[0.04, 0.17]
1.2 Total energy expenditure	8	3376	Std. Mean Difference	0.20
(Kcal/week or Kcal/day)			(IV, Random, 95% CI)	[0.13, 0.27]
1.3 Aerobic fitness VO2 Max	6	8822	Std. Mean Difference	0.06
(ml/kg/min or ml/min)			(IV, Random, 95% CI)	[-0.01, 0.12]
1.4 Body Mass Index	7	2371	Std. Mean Difference	-0.04
(BMI)			(IV, Random, 95% CI)	[-0.15, 0.07]
1.5 Systolic blood pressure	7	2856	Std. Mean Difference	-0.10
(mmHg)			(IV, Fixed, 95% CI)	[-0.18, -0.03]
1.6 Total cholesterol	4	2459	Std. Mean Difference	-0.27
			(IV, Random, 95% CI)	[-0.72, 0.18]
1.8 Waist circumference	4	1469	Std. Mean Difference	-0.05
			(IV, Fixed, 95% CI)	[-0.15, 0.06]

Table 2: Physical activity counselling in primary care setting verses Usual care or placebo

5. Discussion

5.1 Summary of main results

We included 23 randomised controlled trials (RCTs) and cluster randomised trials examining 15,069 patients aged between 18 to 80 years in this review. Sixteen studies examined self-reported physical activity in minutes/week (N =13211 randomised); eight studies examined total energy expenditure (N = 3376 randomised); six studies examined aerobic fitness (N=8822 randomised); seven trials examined body mass index (N=2371 randomised); seven trials examined systolic blood pressure (N=2856 randomised); four trials examined total cholesterol (N=2459 randomised) and four trials examined waist circumference (N = 2385 randomised). Findings suggest that counselling delivered by health providers, probably leads to similar or better physical activity outcomes for patient conditions (moderate-certainty evidence). However, these results must be interpreted with a degree of caution, recognising the variation in interventions reported within studies and the complex interplay of factors affecting outcomes. Several studies included multiple intervention methods, which made it difficult to tease apart which intervention components were the active ingredients. The effectiveness of the interventions was not significant in various outcomes. There was a paucity of data reporting on adverse events.

In summary:

- 1) Counselling delivered by health providers may lead to improved self reported activity, compared to usual care. However, the SMD was small (0.11).
- 2) Total energy expenditure and blood pressure outcomes are probably slightly improved in health provider counselling group.

 There were no observed significant improvements in aerobic fitness, body mass index, total cholesterol and waist circumference An overview can be found in (<u>Summary of findings, Table 1</u>)

5.2 Overall completeness and applicability of evidence

Several issues need to be considered when one is making judgements about the applicability of these findings in primary care setting. First, we were able to identify a moderate number of studies published up to 2014, which were sufficient to address all objectives of the review. These studies are highly varied in terms of types of health providers (physicians and nurses), healthcare setting, and geographical settings, and they examine counselling provided to general patient populations as well as to specific groups of patients, such as people with risk factors for non-communicable diseases. Next, we found a large variation in outcome measures. For a number of outcomes there were only a few contributing studies whereas for some other outcomes a relatively large number of studies contributed to the evidence. Furthermore, often details (such as type of health provider and type of intervention) were missing or unclear from study reports. Therefore, we were not able to conduct planned subgroup analyses. As a result, it is not possible to draw conclusions on the influence of health provider type and different types of the interventions on outcomes. In addition, all included studies were conducted in highincome country settings (Finland, Turkey, New Zealand, Spain, US, Canada, UK and Denmark). In other studies (Aittasalo, Miilunpalo, Harjula & Pasanen, 2006), patients were given an appointment but in others they were only advised to see their physicians (Elley et al., 2003), or included a longer time slot for physical activity consultations (Grandes et al., 2009). These differences in the interventions provided might have influenced study outcomes. Last, over the years, primary care services have changed considerably in many settings. However, we did not identify a trend in types of physical activity counselling interventions or in changes in outcomes assessed that might reflect changes in primary care services.

5.3 Quality of the evidence

We assessed the quality of evidence per outcome using the <u>GRADEpro GDT</u>. All studies had some methodological shortcomings, such as contamination and lack of blinding both patients and personnel, which sometimes led to downgrading of the evidence owing to risk of bias. Although lack of blinding is considered a shortcoming, blinding is often not possible for organizational interventions, such as administering of physical activity counselling and the impact of this on outcomes is unclear. Overall, there was low-quality evidence for outcomes because of high risks of bias, imprecision of effect estimates and inconsistent results. We downgraded our assessments of the quality of evidence produced by the included studies because of risk of bias, lack of blinding and use of subjective outcome measures (detection bias), lack of information on sequence generation (selection bias), and lack of information on allocation concealment (selection bias). Besides the paucity of usable data, the main quality concerns were use of subjective

outcome measures (detection bias), which occurred for all the interventions. Overall, reports were poorly written and, in some cases, it was difficult to follow the analyses, although the more recent studies have improved in their design and analyses.

5.4 Potential biases in the review process

The process of study selection, data extraction, and assessment of risk of bias of included studies was performed by two independent review authors and we resolved disagreements through discussion and consensus. Our search strategy was designed to maximise sensitivity (detecting relevant research) at the expense of specificity (excluding irrelevant research). Even so, relevant research proved difficult to identify, and some studies may have been missed. We conducted this review according to Cochrane standards. Therefore, we are confident in the quality of the review itself. Although publication bias cannot be ruled out in this area, it seems unlikely that this bias could be substantial. We were unable to assess the risk of publication bias adequately as there was a very limited number of studies assessing similar types of interventions and outcomes. We had considerable difficulty in classifying the interventions and we might have been too restrictive in combining studies. However, we believe that the broad categories of physical activity interventions that we made have resulted in a meaningful categorisation. Thus, we believe it is possible to get at least an impression of the effectiveness of interventions in the various outcomes.

5.5 Agreements and disagreements with other studies or reviews

Our results mirrored in other systematic reviews of PA interventions for older adults. A review by Hillsdon, Thorogood & Foster, 2013 found that there was a positive and moderate pooled effect on self-reported PA at 12 months (SMD 0.19; 95% CI 0.06 to 0.31), however the systematic review focused on face to face interventions only. A study by Conn (2011) reported on randomised and non-randomised PA interventions (n = 99,011 participants) and calculated a mean effect size for the comparison of intervention groups versus control groups that was positive but not significant (SMD 0.19; 95% CI -0.14 to 0.53). Hobbs (2013) reported an identical pooled effects size for PA interventions for older adults using counselling and lifestyle advice for self-reported PA outcomes (SMD 0.19; 95% CI 0.10 to 0.28). Foster et al. (2005) found that the effect of interventions on selfreported physical activity (19 studies; 7598 participants) was positive and moderate (pooled SMD random effects model 0.28 95% CI 0.15 to 0.41) as was the effect of interventions (11 studies; 2195 participants) on cardio-respiratory fitness (pooled SMD random effects model 0.52 95% CI 0.14 to 0.90). There was significant heterogeneity in the reported effects as well as heterogeneity in characteristics of the interventions. The heterogeneity in reported effects was reduced in higher quality studies, when physical activity was self-directed with some professional guidance and when there was on-going professional support.

5.6 Authors' conclusions 5.6.1 Implications for practice

The review concludes that to initiate important increases in physical activity, both exercise prescription and counselling should be incorporated in standard practice. We recommend as a tool for primary health care physicians to promote PA, especially at health check and control visits, where more time can be allocated for the appointment. A more comprehensive familiarization protocol or training sessions are needed to implement physical counselling to everyday practices and facilitate counselling cooperation. Self-monitoring of PA with expert feedback can be a useful and cheaper way of increasing especially the duration of overall weekly PA in the short term.

5.6.2 Implications for research

From the review conclusions, comprehensive approaches involving repeated interventions that include behaviour change techniques and booster programs might enhance the long-term effectiveness of advice given by physicians and other health care professionals. Future studies should address the modelling and evaluation of these new and complex interventions and also demonstrate what intensity of follow-up is needed for longer-term maintenance

Acknowledgements

The authors wish to acknowledge Prof Edwin Kadima Wamukoya, and Dr Maximilla Wanzala for their contributions and support during the writing of this review.

Contributions of authors

Micky Olutende Oloo, Prof Edwin Wamukoya and Dr. Maximilla Wanzala conceived the paper, designed and performed the review. All authors read and approved the final manuscript

Declarations of interest

The authors declare that they have no competing interests. The findings and conclusions presented in this manuscript are those of the authors and do not necessarily reflect the official position of Masinde Muliro University of Science and Technology

References

- Aittasalo, M., Seppo, M., Katriina, K., Matti P. (2006). A randomised intervention of physical activity promotion and patient self-monitoring in primary health care. *Preventive Medicine*, (42), 40-46.
- Emrah A., Naciye, F. T., Hakan, Y. (2014). Exercise Prescription by Primary Care doctors: Effect on Physical Activity level and Functional Abilities in Elderly. *Turkish Journal of Geriatrics*, 17(1), 77-85.

- Elley, C., Kerse, N., Arroll, B., & Robinson, E. (2003). Effectiveness of counselling patients on physical activity in general practice: cluster randomised controlled trial. British Medical Journal, (326), 793-799. [DOI: 10.1136/bmj.326.7393.793]
- Michelle S. F., William H., Tracey L., O'Sullivan, C. B., Ronald J. S., Robert D. R., Pierre, B., Éric D., Étienne B., Julie, B., and Diane, C. (2011). Impact of integrating a physical activity counsellor into the primary health care team: physical activity and health outcomes of the Physical Activity Counselling randomised controlled trial. *Appl. Physiol. Nutr*, 36.
- Goldstein, M. G., Pinto, B. M., Marcus, B. H., Lynn, H., Jette, A. M., Rakowski, W., ... Tennstedt, S. (1999). Physician-based physical activity counselling for middleaged and older adults: A randomized trial. Annals of Behavioral Medicine, (21), 40-47. [DOI: 10.1007/BF02895032]
- Grandes, G., Sanchez, A., Sanchez-Pinilla, R. O., Ortega, R., Torcal, J., Montoya, I., ... Serra, J. (2009). Effectiveness of physical activity advice and prescription by physicians in routine primary care. A cluster randomized trial. *Archives of Internal Medicine*, 169(7), 694-701. [DOI: 10.1001/archinternmed.2009.23]
- Gonzal, G., Alvaro, S., Imanol, M., Ricardo, O., Sanchez, P., Jesus, T.(2011). Two-year Longitudinal Analysis of a cluster randomised Trial of Physical Activity Promotion by General Practitioners. *Plos One*, 6(3).
- Beverly, B.G., Timothy, M., Michael, H., Linda, M., Madlen, C., Diana, Buist.(2002). Effectiveness of Telephone Support in Increasing Physical Activity Levels in Primary Care Patients. American Journal of Preventive Medicine, 22(3), 177-183.
- Melyvyn, H., Margaret, T., Ian, W. & Charlie, F. (2002). Advising people to take more exercise is ineffective: a randomised controlled trial of physical activity promotion in primary care. international journal of epidemiology, *31*, 808-815.
- Kerse, N., Elley, C. R., Robinson, E., Arroll, B. (2005). Is physical activity counselling effective for older people? A cluster randomised, controlled trial in primary care. *Journal of the American Geriatric Society*, 53(11), 1951-1956.
- Gregory, S. K., Grant, M. S., Ngaire, K., Nick, G., and Melody, O. (2007). Effect of Telephone Counseling on Physical Activity for Low-Active Older People in Primary Care: A Randomized, Controlled Trial. JAGS, 55, 986-992.
- Beverley, A. L., Sally, B. R., Raina, E., Anthony, C. D., Anna, F., Simon, A. M. (2008). Exercise on prescription for women aged 40-74 recruited through primary care: two year randomised controlled trial. *BMJ*, 337.
- Lewis, B. S, & Lynch, W. D. (1993). The effect of physician advice on exercise behavior. *Preventive medicine*, 22, 110-121.
- Little, P., Dorward, M., Gralton, S., Hammerton, L., Pillinger, J., & White, P. (2004). A randomised controlled trial of three pragmatic approaches to initiate increased physical activity in sedentary patients with risk factors for cardiovascular disease. *British Journal of General Practice*, 54(500), 189-195.
- Mutriea, N., Orla D., Fitzsimonsa, C. F., Margaret G.P., Malcolm, G., Madeleine, G., Hazel, M., Freya, M., McConnachieb, A., David A. R., Rebecca, S. and Dawn, A. S.

(2014). Increasing older adults' walking through primary care:results of a pilot randomised controlled trial. *Family Practice*, (29) 633-642. [DOI: 10.1093/fampra/cms038]

- Norris, S. L., Grothaus, L. C., Bucner, D. M., & Pratt, M. (2000). Effectiveness of physicianbased assessment and counselling for exercise in a staff model HMO. *Preventive Medicine*, 30, 513-523. [DOI: 10.1006/pmed.2000.0673]
- Sally, P., Maaike, B., Katie, M., Joana, V., Richard, A., Parker, K.W., Soren, B., Wilson, E., Toby P., Ann-Louise, K., Simon, G., Stephen, S., Hardeman. W. (2016). A randomised controlled trial of three very brief interventions for physical activity in primary care. *BMC Public Health*, 16.
- Robert, J., Petrella, C. N., Lattanzio, S. S., Tom, O. (2010). Improving aerobic fitness in older adults. *Canadian Family Physician*, 56, 191-200.
- Petrella, R. J., Koval, J. J., Cunninham, D. A., & Paterson, D. H.(2003). Can primary care doctors prescribe exercise to improve fitness? The step test exercise prescription (STEP) project. *American Journal of Preventive Medicine*, 24, 316-322. [DOI: 10.1016/S0749-3797(03)00022-9]
- Pfeiffer, B. A., Clay, S. W., & Conatser Jr., R. R. (2001). A green prescription study: Does written exercise prescribed by a physician result in increased physical activity among older adults? *Journal of Aging and Health*, 13:527-538. [DOI: 10.1177/089826430101300405]
- Elizabeth, L.R., & Robert W. M. (1979). Exercise Prescription: A Clinical Trial. *Am. J.Public Health*, 69, 591-595.
- Swinburn, B. A., Walter, L. G., Arroll, B., Tilyard, M. W., & Russell, D. G. (1998). The Green Prescription Study: A randomised controlled trial of written exercise advice provided by general practitioners. *American Journal of Public Health*, 88, 288-291. [DOI: 10.2105/AJPH.88.2.288]
- Sørensen, J., Kragstrup, T., Skovgaard, L., Puggaard. (2008). Exercise on prescription: a randomised study on the effect of counselling vs counselling and supervised exercise. *Scand J Med Sci Sports*, 18, 288-297. [DOI: 10.1111/j.1600-0838.2008.00811.x]
- Taylor, A. H. & Fox, K. R. (2005). Effectiveness of a Primary Care Exercise Referral Intervention for Changing Physical Self-Perceptions Over 9 Months. *Health Psychology*, 24(1), 11-21.

A. References to studies

a. Included studies

Aittasalo, Miilunpalo, Harjula & Pasanen (2006)

Aittasalo, M., Seppo, M., Katriina, K., Matti P. (2006). A randomised intervention of physical activity promotion and patient self-monitoring in primary health care. *Preventive Medicine*, (42), 40-46.

Atay, Torman & Yaman (2014)

Emrah A., Naciye, F.T., Hakan, Y. (2014). Exercise Prescription by Primary Care doctors: Effect on Physical Activity level and Functional Abilities in Elderly. *Turkish Journal of Geriatrics*, 17(1), 77-85.

Elley et al. (2003)

Elley, C., Kerse, N., Arroll, B., & Robinson, E. (2003). Effectiveness of counselling patients on physical activity in general practice: cluster randomised controlled trial. British Medical Journal, (326), 793-799. [DOI: 10.1136/bmj.326.7393.793]

Fortier et al. (2011)

Michelle S. F., William H., Tracey L., O'Sullivan, C. B., Ronald J. S., Robert D. R., Pierre, B., Éric D., Étienne B., Julie, B., and Diane, C. (2011). Impact of integrating a physical activity counsellor into the primary health care team: physical activity and health outcomes of the Physical Activity Counselling randomised controlled trial. *Appl. Physiol. Nutr*, 36.

Goldstein et al. (1999)

Goldstein, M. G., Pinto, B. M., Marcus, B. H., Lynn, H., Jette, A. M., Rakowski, W., ... Tennstedt, S. (1999). Physician-based physical activity counselling for middle-aged and older adults: A randomized trial. Annals of Behavioral Medicine, (21), 40-47. [DOI: 10.1007/BF02895032]

Grandes et al. (2009)

Grandes, G., Sanchez, A., Sanchez-Pinilla, R. O., Ortega, R., Torcal, J., Montoya, I., ... Serra, J. (2009). Effectiveness of physical activity advice and prescription by physicians in routine primary care. A cluster randomized trial. *Archives of Internal Medicine*, 169(7), 694-701. [DOI: 10.1001/archinternmed.2009.23]

Grandes et al. (2011)

Gonzal, G., Alvaro, S., Imanol, M., Ricardo, O., Sanchez, P., Jesus, T.(2011). Two-year Longitudinal Analysis of a cluster randomised Trial of Physical Activity Promotion by General Practitioners. *Plos One*, 6(3).

Green et al. (2002)

Beverly, B.G., Timothy, M., Michael, H., Linda, M., Madlen, C., Diana, Buist.(2002). Effectiveness of Telephone Support in Increasing Physical Activity Levels in Primary Care Patients. American Journal of Preventive Medicine, 22(3), 177-183.

Hillsdon et al. (2002)

Melyvyn, H., Margaret, T., Ian, W. & Charlie, F. (2002). Advising people to take more exercise is ineffective: a randomised controlled trial of physical activity promotion in primary care. international journal of epidemiology, *31*, 808-815.

Kerse et al. (2005)

Kerse, N., Elley, C. R., Robinson, E., Arroll, B. (2005). Is physical activity counselling effective for older people? A cluster randomised, controlled trial in primary care. *Journal of the American Geriatric Society*, 53(11), 1951-1956.

Kolt et al. (2007)

Gregory, S. K., Grant, M. S., Ngaire, K., Nick, G., and Melody, O. (2007). Effect of Telephone Counseling on Physical Activity for Low-Active Older People in Primary Care: A Randomized, Controlled Trial. *JAGS*, 55, 986-992.

Lawton et al. (2008)

Beverley, A. L., Sally, B. R., Raina, E., Anthony, C. D., Anna, F., Simon, A. M. (2008). Exercise on prescription for women aged 40-74 recruited through primary care: two year randomised controlled trial. *BMJ*, 337.

Lewis & Lynch (1993)

Lewis, B. S, & Lynch, W. D. (1993). The effect of physician advice on exercise behavior. *Preventive medicine*, 22, 110-121.

Little et al. (2004)

Little, P., Dorward, M., Gralton, S., Hammerton, L., Pillinger, J., & White, P. (2004). A randomised controlled trial of three pragmatic approaches to initiate increased physical activity in sedentary patients with risk factors for cardiovascular disease. *British Journal of General Practice*, 54(500), 189-195.

Mutrie et al. (2012)

Mutriea, N., Orla D., Fitzsimonsa, C. F., Margaret G.P., Malcolm, G., Madeleine, G., Hazel, M., Freya, M., McConnachieb, A., David A. R., Rebecca, S. and Dawn, A. S. (2014). Increasing older adults' walking through primary care:results of a pilot randomised controlled trial. *Family Practice*, (29) 633-642. [DOI: 10.1093/fampra/cms038]

Norris et al. (2000)

Norris, S. L., Grothaus, L. C., Bucner, D. M., & Pratt, M. (2000). Effectiveness of physicianbased assessment and counselling for exercise in a staff model HMO. *Preventive Medicine*, 30, 513-523. [DOI: 10.1006/pmed.2000.0673]

Pears et al. (2016)

Sally, P., Maaike, B., Katie, M., Joana, V., Richard, A., Parker, K.W., Soren, B., Wilson, E., Toby P., Ann-Louise, K., Simon, G., Stephen, S., Hardeman. W. (2016). A randomised controlled trial of three very brief interventions for physical activity in primary care. *BMC Public Health*, 16.

Petrella, Lattanzio, Shapiro & Overend (2010)

Robert, J., Petrella, C. N., Lattanzio, S. S., Tom, O. (2010). Improving aerobic fitness in older adults. *Canadian Family Physician*, 56, 191-200.

Petrella et al. (2003)

Petrella, R. J., Koval, J. J., Cunninham, D. A., & Paterson, D. H.(2003). Can primary care doctors prescribe exercise to improve fitness? The step test exercise prescription (STEP) project. *American Journal of Preventive Medicine*, 24, 316-322. [DOI: 10.1016/S0749-3797(03)00022-9]

Pfeiffer et al. (2001)

Pfeiffer, B. A., Clay, S. W., & Conatser Jr., R. R. (2001). A green prescription study: Does written exercise prescribed by a physician result in increased physical activity among older adults? *Journal of Aging and Health*, 13:527-538. [DOI: 10.1177/089826430101300405]

Reid & Morgan (1979)

Elizabeth, L.R., & Robert W. M. (1979). Exercise Prescription: A Clinical Trial. *Am. J.Public Health*, 69, 591-595.

Swinburn et al (1998)

Swinburn, B. A., Walter, L. G., Arroll, B., Tilyard, M. W., & Russell, D. G. (1998). The Green Prescription Study: A randomised controlled trial of written exercise advice provided by general practitioners. *American Journal of Public Health*, 88, 288-291. [DOI: 10.2105/AJPH.88.2.288]

Sørensen et at. (2008)

Sørensen, J., Kragstrup, T., Skovgaard, L., Puggaard. (2008). Exercise on prescription: a randomised study on the effect of counselling vs counselling and supervised exercise. *Scand J Med Sci Sports*, 18, 288-297. [DOI: 10.1111/j.1600-0838.2008.00811.x]
Tylor & Fox (2005)

Taylor, A. H. & Fox, K. R. (2005). Effectiveness of a Primary Care Exercise Referral Intervention for Changing Physical Self-Perceptions Over 9 Months. *Health Psychology*, 24(1), 11-21.

b. Excluded studies

Activity (2001)

The writing Group for the activity counselling trial research group. (2001) Effects of Physical activity counselling in Primary care. JAMA, 286(6), 677-687.

Armit et al. (2009)

Armit, C. M, Wendy J. B., Alison, L. M., Carrie, B. R., Stewart, G. T., Anita, G., Adrian, E. B. (2009). Randomized trial of three strategies to promote physical activity in general practice. *Preventive Medicine*,48, 156-163.

Bennet et al. (2008)

Jill A., Bennett, H. M., Young, L.M., Nail, K.W., Ginger, H. (2008). A Telephone-Only Motivational Intervention to Increase Physical Activity in Rural Adults. *Nursing Research*, 57(1):24-32.

Bucholz & Purath (2007)

Buchholz, S. W., & Purath, J. (2007). Physical activity and physical fitness counselling patterns of adult nurse practitioners. *Journal of the American Academy of Nurse Practitioner*, 19, 86-92. [DOI: 10.1111/j.1745-7599.2006. 00197.x]

Bull & Jamrozik (1999)

Bull, F. C., & Jamrozik, K. (1999). Advice on exercise from a family physician can help sedentary patients to become active. *American Journal of Preventive Medicine*, 15(2), 85-94. [DOI: 10.1016/S0749-3797(98)00040-3]

Burn, Camaione & Chartterton (2000)

Burns, K. J., Camaione, D. N., & Chatterton, C. T. (2000). Prescription of physical activity by adult nurse practitioners: A national survey. *Nursing Outlook*, 48, 28-33. [DOI: 10.1067/mno.2000.99101]

Calfas et al. (1996)

Calfas, K. J., Long, B. J., Sallis, J. F., Wooten, W. J., Pratt, M., & Patrick, K. (1996). A controlled trial of physician counselling to promote the adoption of physical activity. *Preventive Medicine*, 25, 225-233. [DOI: 10.1006/pmed.1996.0050]

Cunningham et al. (1987)

Cunningham, D. A., Rechnitzer, P. A., Howard, J. H. and Donner, A. P. (1987). Exercise Training of Men at Retirement: *A Clinical Trial. Journal of Gerontology* 42, (1), 17-23.

Dubbert et al. (2002)

Patricia, M. D., Karen, M.C., Kent A. K., Edward, F. M., and Deborah, B. (2002). Effects of Nurse Counseling on Walking for Exercise in Elderly Primary Care Patients. *Journal of Gerontology*, 11, 733-740.

Duru et at. (2010)

Duru, O. K., Catherine A. S., Mei L., and Carol, M. M. (2010). Sisters in Motion: A Randomized Controlled Trial of a Faith-Based Physical Activity Intervention. *JAGS*, 58, 1863-1869.

Fortier et al. (2006)

Fortier, M.S., William, H., Tracey, L. O., Christopher, B., Robert, D., Reid, R.J., Sigal et al. (2007). The physical activity counselling (PAC) randomised controlled trial: rationale, methods and interventions. *Appl. Physiol. Nutr*, 32.

Gao et al. (2016)

Gao, S., Roslyn A. S., Linda J. H., Jeffrey, P. H., Bess, H. M., Joseph, T. C., Andrea, M. K., Kelly, H. B., Ann, R. S., Marie, A. B., Mary, A. S. (2016). Physical activity counselling in overweight and obese primary care patients: Outcomes of the VA-STRIDE randomised controlled trial. *Preventive Medicine Reports*, 3, 11-120.

Hardcastle (2012)

Sarah, H., Nicola, B., Martin S. H. (2012). The effectiveness of a motivational interviewing primary-care based intervention on physical activity and predictors of change in a disadvantaged community. *J Behav Med*, 35, 318-333. [DOI: 10.1007/s10865-012-9417-1]

Harland et al. (1999)

Harland, J., White, M., Drinkwater, C., Chinn, D., Farr, L., & Howel, D. The Newcastle exercise project: A randomised controlled trial of methods to promote physical activity in primary care. British Medical Journal 1999;319(7213):828-832.

Harrison, Roberts & Elton (2004)

Roger A. H., Chris, R. & Peter, J. (2004). Elton. Does primary care referral to an exercise programme increase physical activity 1 year later? A randomised controlled trial. *Journal of Public Health*, 27(1), 25-32. [DOI: 10.1093/pubmed/fdh197]

Jimmy & Martin (2005)

Gerda, J. & Martin, B.W. (2006). Implementation and effectiveness of a primary care based physical activity counselling scheme. *Patient Education and Counseling*, 56, 323-331.

Kastarinen et al. (2002)

Kastarinen, M. J., Puska, P. M., Korhone, M. H., Mustonen, J. N., Salomaa, V. V., & Sundvall, J. E. (2002). Non-pharmacological treatment of hypertension in primary health care: A 2-year open randomised controlled trial of lifestyle intervention against hypertension in eastern Finland. *Journal of Hypertension*, 20(12), 2505-2512. [DOI: 10.1097/00004872-200212000-00031]

Kriska et al. (1986)

Kriska, A. M., Connie B., Jane A. C., Ronald, E. L., Rivka, B.S. & Georgia, P. (1986). A randomised exercise trial in older women: Increased activity over two years and the factors associated with compliance. *Medicine and science in sports and exercise*, 557-562.

Long et al. (1996)

Long, B. J., Calfas, K. J., Wooten, W., Sallis, J. F., Patrick, K., Goldstein, M., … Heath, G. (1996). A multisite field test of the acceptability of physical activity counselling in primary care: Project PACE. *American Journal of Preventive Medicine*, 25, 225-235.

Marcus et al. (2007)

Bess, H. M., Melissa, A. N., Abby, C. K., Beth, A. L., Jessica, A. W., Anna, E. A., Alfred, F. P., Beth, C. B., Bernardine, M. P., Christopher, A. S., John, M. J., George, D. P.(2007). Examination of print and telephone channels for physical activity promotion: Rationale, design, and baseline data from Project STRIDE. *Contemporary Clinical Trials*, *28*,90-104.

Nymberg et al. (2018)

Peter, N., Eva, E. H., Emelie, S., Susanna, C., Kristina, S., Jan, S. & Bengt, Z. (2018). Pilot study on increased adherence to physical activity on prescription (PAP)through mindfulness: study protocol. *BMC*, 19. [DOI: 10.1186/s13063-018-2932-9]

Pekmezi et al. (2016)

Pekmezi, D., Cole, A., Rodney, J., Molly S. B., Elizabeth K., Shiney, I., Renee, D., Karen, M., Bess, M., and Wendy, D. (2016). Rationale, design, and baseline findings from HIPP: A randomised controlled trial testing a Home-based, Individually-tailored Physical activity Print intervention for African American women in the Deep South. *Contemp Clin Trials*, *47*, 340-348. [DOI: 10.1016/j.cct.2016.02.009]

Pinto et al. (1998)

Pinto, B. M., Goldstein, M. G., DePue, J. D., & Milan, F. B. (1998). Acceptability and feasibility of physician-based activity counselling. The PAL projects. *American Journal of Preventive Medicine*, 15, 95-102. [DOI: 10.1016/S0749-3797(98)00043-9]

Rome, Ekdahl & Gard (2009)

Romé, A., Ulf, P., Charlotte, E. & Gunvor, G. (2009). Physical activity on prescription (PAP): Costs and consequences of a randomised, controlled trial in primary healthcare. *Scandinavian Journal of Primary Health Care*, 27, 216-222.

Schröder et al. (2018)

Schröder, G. C., Miguel, A.G., Dolores, C.J.V., Dora, R., Alfredo, M. et al. (2018) Effectiveness of the physical activity intervention program in the PREDIMED-Plus study: a randomised controlled trial. *International Journal of Behavioral Nutrition and Physical Activity*, 15. [DOI: 10.1186/s12966-018-0741-x]

Skår et al. (2011)

Skår, S., Falko F. S., Gerard J. M., Andrew, P. & Vera, A.S. (2011). Do brief online planning interventions increase physical activity amongst university students? A randomised controlled trial, 26(4):399-417.

Spittaels, Bourdeaudhuij & Vandelanotte (2007)

Spittaels, H., Ilse D. B., Corneel, V. (2007). Evaluation of a website delivered computertailored intervention for increasing physical activity in the general population. *Preventive Medicine*, 44, 209-217.

Steele, Mummery & Dwyer (2009)

Rebekah, M. S., Mummery, T.D. (2009). A Comparison of Face-to-Face or Internet-Delivered Physical Activity Intervention on Targeted Determinants. *Health Education & Behavior*, 36(6). [DOI: 10.1177/1090198109335802]

Stralen et al. (2009)

Stralen, M. M., Hein, V., Aart, N. M., Catherine, B., Lilian, L. (2009). Efficacy of Two Tailored Interventions Promoting Physical Activity in Older Adults. *Am J Prev Med*, 37(5), 405-417.

Wanner et al. (2009)

Miriam, W., Eva, M.D., Charlotte, B.F., Georg, B., Brian, W. M. (2009). Effectiveness of Active-Online, an Individually Tailored Physical Activity Intervention, in a Real-Life Setting: Randomized Controlled Trial. *Journal of Medical Internet Research*, 11(3).

William et al. (2006)

David M., Williams, G.D., Papandonatos, M.A., Napolitano, B.A., Lewis, J. A., Whiteley, and Bess, H. M. (2006). Perceived Enjoyment Moderates the Efficacy of an Individually Tailored Physical Activity Intervention. *Journal of Sport & Exercise Psychology*, 28, 300-309.

c. Studies awaiting classification

Ongoing studies

Missud et al. (2019)

David, C., Missud, E.P., Laurent, C., Bruno, V. & Jean-François, H. (2019). Physical activity prescription for general practice patients with cardiovascular risk factors–the PEPPER randomised controlled trial protocol. *BMC Public Health*.

Rose et al. (2007)

Sally, B., Rose, B. A., Lawton, C., Raina, E., Anthony, C., Dowell and Anna J. F. (2007). The 'Women's Lifestyle Study', 2-year randomised controlled trial of physical activity counselling in primary health care: rationale and study design. *BMC Public Health*, 7.

C. Other references

a. Additional references

Bartholomew et al. (2001)

Bartholomew L. K., Parcel G. S., Kok G., Gottlieb N. H. (2001) Intervention mapping: designing theory- and evidence-based health promotion programs. Mountain View: Mayfield.

Bauman et al. (2012)

Bauman, A.E., Reis, R.S., Sallis, J.F., Wells, J.C., Loos, R.J., Martin, B.W. (2012). Correlates of physical activity: why are some people physically active and others not? Lancet 380(9838), 258-271.

Biddle & Mutrie (2001)

Biddle, S., Mutrie, N. (2001). Psychology of physical activity: determinants, well-being and interventions. London: Routledge.

Cohen (1988)

Cohen, J. (1988). Statistical Power Analysis for the Behavioral Sciences. 1st edition. New York, San Francisco: London: Academic press.

Conn (2011)

Conn, V., Hafdahl A., Mehr D. (2011). Interventions to increase physical activity among healthy adults: Meta-analysis of outcomes. *American Journal of Public Health*, 101(4), 751-758. [DOI: 10.2105/AJPH.2010.194381]

Das & Horton (2016)

Das, P. & Horton, R. (2016). PA???time to take it seriously and regularly. The Lancet, 388(10051), 1254-1255.

Derman et al. (2008)

Derman, E., Patel, D. N., Nossel, C. J. & Schellnus, M. P. (2008). Healthy lifestyle interventions in general practice Part 1: An introduction to lifestyle and diseases of lifestyle. *South African Family Practice*, 50(4), 19-22.

Ding et al. (2016)

Ding, D., Lawson, K. D., Kolbe-Alexander, T. L., Finkelstein, E. A., Katzmarzyk, P. T., van Mechelen, W. & Pratt, M. (2016). The economic burden of physical inactivity: a global analysis of major non-communicable diseases. *The Lancet*, 388(10051), 1311-1324.

Fiuza-Luces et al. (2013)

Fiuza-Luces, C., Garatachea, N., Berger, N. A. & Lucia, A. (2013). Exercise is the real polypill. Physiology, 28(5), 330-358.

Foster et al. (2005)

Charles, F., Melvyn, H., Margaret, T., Asha, K., Thamindu, W. (2005). Interventions for promoting physical activity. *Cochrane Database of Systematic Reviews*. [DOI: 10.1002/14651858.CD003180.pub2.]

Garry, Diamond & Whitley (2002)

Garry, J. P., Diamond, J. J., and Whitley, T. W. (2002). Physical activity curricula in medical schools. *Academic Medicine*, 77, 818-821. [DOI: 10.1097/00001888-200208000-00011]

GRADEpro GDT

GRADEproGDT [Computer program]. (2015). McMaster University (developed by Evidence Prime). Version accessed November 2019.

Green et al. (2011)

Green, D. J., Spence, A., Halliwill, J. R., Cable, N. T. & Thijssen, D. H. (2011). Exercise and vascular adaptation in asymptomatic humans. Experimental physiology, 96(2), 57-70.

Grimstvedt et al. (2013)

Grimstvedt, M., Der Ananian, C., Pasquariello, C. D., Masambia, F. M., Chevan, J., Haskvitz, E. M., ...& Rogers, L. Q. (2013). The role of professional counselling associations in professional counselling practice in Kenya. *Medicine and Science in Sports and Exercise*, 77(3), 200. [DOI: 10.2105/AJPH.69.6.591]

Hallal (2012)

Hallal, P. C., Andersen, L. B., Bull, F. C., Guthold, R., Haskell, W., Ekelund, U. (2012). Global physical activity levels: surveillance progress, pitfalls, and prospects. *Lancet*, 380(9838):247-257.

Heath et al. (2012)

Heath, G. W., Parra, D. C., & Sarmiento, O.L. (2012). Evidence-based intervention in PA: lessons from around the world. Lancet, 380, 272-281. [PubMed: 22818939]

Higgins (2011)

Higgins, J. P. T., Green, S. (2011). Cochrane Handbook for Systematic Reviews of Interventions Version 5.1.0. The Cochrane Collaboration. [Other: <u>www.handbook.cochrane.org</u>.]

HMIS (2012)

Kenya Health Management Information system. [MOH]. 2012.

Hobbs (2013)

Hobbs N., Godfrey A., Lara L., Errington L., Meyer T., Rochester L., et al. (2013) Are behavioral interventions effective in increasing physical activity at 12 to 36 months in adults aged 55 to 70 years? A systematic review and meta-analysis. *BMC Medicine*, 11. [DOI: 10.1186/1741-7015-11-75]

Hébert, Caughy & Shuval (2012)

Hébert, E. T., Caughy, M. O., Shuval, K. (2012). Primary care providers' perceptions of physical activity counselling in a clinical setting: a systematic review. *Br J Sports Med*, 46(9), 625-631.

Jacobson et al. (2005)

Jacobson, D. M., Strohecker, L., Compton, M. T., and Katz, D. L. (2005). Physical activity counselling in the adult primary care setting. Position statement of the American College of Preventive Medicine. *American Journal of Preventive*, 29, 158-162. [DOI: 10.1016/j.amepre.2005.04.009]

Kenya Demographic and Health Survey (2014)

Kenya National Bureau of Statistics, & ICF Macro. Kenya Demographic and Health Survey. 2014.

Kesaniemi et al. (2001)

Kesaniemi Y. K., Danforth E. Jr., Jensen M. D., Kopelman P. G., Lefebvre P., Reeder B. A. (2001). Dose-response issues concerning physical activity and health: an evidence-based symposium. *Medicine and Science in Sports and Exercise*, 33(6), 351-358.

Kimmel, Haas, & Hermanns (2014)

Kimmel, G. T., Haas, B. K. & Hermanns, M. (2014). The Role of Exercise in Cancer Treatment: Bridging the Gap. *Current Sports Medicine Reports*, 13(4), 246-252. [DOI: 10.1249/JSR.000000000000068]

Lamming et al. (2017)

Lamming, L., Pears, S., Mason, D., Morton, K., Bijker, M., Sutton, S. & Hardeman, W. (2017). What do we know about brief interventions for PA that could be delivered in primary care consultations? A systematic review of reviews. *Preventive Medicine*, 99, 152-163. [DOI: 10.1016/j.ypmed.2017.02.017]

Lanhers et al. (2015)

Lanhers, C., Duclos, M., Guttmann, A., Coudeyre, E., Pereira, B. & Ouchchane, L. (2015). General practitioners' barriers to prescribe PA: The dark side of the cluster effects on the PA of their type 2 diabetes patients. *PLoS ONE*, 10(10), 1-12. [DOI: 10.1371/journal.pone.0140429]

Lee et al. (2012)

Lee I. M., Shiroma E. J., Lobelo F., Puska P., Blair S. N., Katzmarzyk P. T. (2012). Effect of physical inactivity on major non-communicable diseases worldwide: an analysis of burden of disease and life expectancy. *Lancet*, 380(9838), 219-229.

Leemrijse et al. (2015)

Leemrijse, C. J., de Bakker, D. H., Ooms, L. & Veenhof, C. (2015). Collaboration of general practitioners and exercise providers in promotion of PA a written survey among general practitioners. *BMC Family Practice*, 16(1):96. [DOI: 10.1186/s12875-015-0316-8]

Lefebvre (2011)

Lefebvre, C., Manheimer, E., & Glanville, J. (2011). Chapter 6: Searching for studies. In: Higgins JPT, Green S (editors). Cochrane Handbook for Systematic Reviews of Interventions. *The Cochrane Collaboration* 2011. [Other: <u>www.handbook.cochrane.org</u>]

Lobelo & de Quevedo (2016)

Lobelo, F. & de Quevedo I. G. (2016). The Evidence in Support of Physicians and Health Care Providers as PA Role Models. American Journal of Lifestyle Medicine 2016;10(1):36-52. [DOI: 10.1177/1559827613520120]

Machio (2012)

Machio, P. M. (2010). The Effect of Chronic Illness on Labor Market Outcomes in Kenya. Center for the Study of Africa Economies 2010. [Other: Economies. Oxford, UK. <u>https://editorialexpress.com/cgi-</u>

bin/conference/download.cgi?db_name=CSAE2012&paper_id=502]

Matheson et al. (2011)

Matheson, G. O., Klugl, M., Dvorak, J., Engebretsen, L., Meeuwisse, W. H., Schwellnus, M., ... & Weiler, R.(2011). Responsibility of sport and exercise medicine in preventing and managing chronic disease: applying our knowledge and skill is overdue. *Journal of Sports Medicine*, 45(16):1272-1282. [DOI: 10.1136/bjsports-2011-090328]

Naci & Ioannidis (2013)

Naci, H. & Ioannidis, J. P. A. (2013). Comparative effectiveness of exercise and drug interventions on mortality outcomes: meta-epidemiological study. *BMJ*, 344. [DOI: 10.1136/bmj.f5577]

O'Brien et al. (2017)

O'Brien, M. W., Shields, C. A., Oh, P. I. & Fowles, J. R.(2017). Health care provider confidence and exercise prescription practices of Exercise is Medicine Canada workshop attendees. *Applied Physiology, Nutrition, and Metabolism,* 42(4):384-390. [DOI: 10.1139/apnm-2016-0413]

Review Manager 5.3

The Cochrane Collaboration. Review Manager (RevMan). [Computer program].(2014). The Nordic Cochrane Centre. Version 5.3.

Richards, Hillsdon, Thorogood & Foster (2013)

Richards J., Hillsdon M., Thorogood M., Foster C.(2013). Face-to-face interventions for promoting physical activity. *Cochrane Database of systematic Reviews*, (9). [DOI: 10.1002/14651858.CD010392.pub2]

Rodgers & Brawley (1991)

Rodgers, W., Brawley, L. R.(1991). The role of outcome expectancies in participation motivation.. *Journal of Sport & Exercise Psychology*, 13(4), 411-427.

Sallis & Owen (1997)

Sallis, J. F., Owen, N. (1997). Health behaviour and health education: theory, research, and practice. San Francisco. Jossey-Bass.

Sallis (2009)

Sallis J. F. (2009). Exercise is medicine and physicians need to prescribe it! *British Journal of Sports Medicine*, 43, 3-4. [DOI: 10.1136/bjsm.2008.054825]

Sallis (2011)

Sallis, R.(2011). Developing healthcare systems to support exercise: exercise as the fifth vital sign. *British Journal of Sports Medicine* 45, 473-474.

Soares-Miranda et al. (2016)

Soares-Miranda, L., Siscovick, D. S., Psaty, B. M., Longstreth, W. & Mozaffarian, D. (2016). PA and Risk of Coronary Heart Disease and Stroke in Older Adults: The Cardiovascular Health Study HHS Public Access. *Circulation*, 12(1332), 147-155. [DOI: 10.1161/Circulationaha.115.018323]

Sterne (2001)

Sterne, J. A. C., Egger, M., Smith J. D. (2001). Investigating and dealing with publication and other biases in meta-analysis. *British Medical Journal*, 323,101-105.

Teferi, Kumar & Singh (2017)

Teferi, G., Kumar, H. & Singh, P. (2017). PA Prescription for Non-Communicable Diseases: Practices of HCPs in Hospital Setting, Ethiopia. *IOSR Journal of Sports and Physical Education*, 4(1), 54-60. [DOI: 10.9790/6737-04015460]

Tero et al. (2017)

Tero, P., Samppa, T., Ari, H. & Urho, M. K. (2016). Exercise therapy for functional capacity in chronic diseases: an overview of meta-analyses of randomised controlled trials. *British Journal of Sports Medicine*. [DOI: 10.1136/bjsports-2016-097132]

Thent, Das & Henry (2013)

Thent, Z. C., Das, S. & Henry, L. J.(2013). Role of exercise in the management of diabetes mellitus: The global scenario. *PLoS ONE*, 8(11), 1-8. [DOI: 10.1371/journal.pone.0080436]

Tulloch, Fortier & Hogg (2006)

Tulloch H., Fortier M., and Hogg W. (2006). Physical activity counselling in primary care: Who has and who should be counselling? Patient Education and Counseling, 64, 6-20. [DOI: 10.1016/j.pec.2005.10.010]

Vuori et al. (2013)

Vuori, I. M., Lavie, C. J. & Blair, S. N. (2013). PA Promotion in the Health Care System. Mayo Clin Proceedings, 88(12), 1446-1461.

Whitlock, Orleans, Pender & Allan (2002)

Whitlock, E. P., Orleans, T., Pender, N., and Allan, J. (2002). Evaluating primary care behavioral counselling interventions. An evidence-based approach. *American Journal of preventive medicine*, 22, 267-284. [DOI: 10.1016/S0749-3797(02)00415-4]

WHO (2010)

World Health Organization (WHO). Global recommendations on physical activity for health. Geneva. 2010.

WHO (2017)

World Health Organization. Preventing Noncommunicable Diseases (NCDs) Reducing Environmental Risk Factors [WHO]. 2017.

Wiklund (2016)

Wiklund, P. (2016). The role of PA and exercise in obesity and weight management: Time for critical appraisal. *Journal of Sport and Health Science*, 5(2), 151-154. [DOI: 10.1016/j.jshs.2016.04.001]

b. Other published versions of this review

Oloo, Kweyu & Ashiali (2017)

Oloo M. O., Kweyu I. W., Ashiali I.V. (2017). Review of randomised primary care physical activity counselling interventions. *International Journal of Current Research*, Issue 8.

c. Sources of support

Internal sources

• No sources of support provided

External sources

• No sources of support provided

B. Characteristics of studies

a. Characteristics of included studies

1. Aittasalo, Miilunpalo, Harjula & Pasanen (2006)

Methods	The study was a cluster randomised trial conducted in Finland during 2003–2004.								
Participants	Physicians from 24 health care units (N = 67) were randomised to a prescription or a								
	non-prescription group. The patients (N = 265) were assigned to the groups according								
	to their physician. Every other patient of the non-prescription physicians received a								
	pedometer and a physical activity log (MON) and feedback about their 5-day-								
	recordings, the rest served as controls (CON). PA was assessed prior and 2 and 6								
	months after the physician's appointment with a questionnaire. All 20 – 65-year-old								
	patients who made an appointment with the participating physicians were asked to								
	participate. The inclusion criteria were 30 min of moderate-intensity PA on fewer than								
	4 days weekly and no perceived obstacles for PA								
Interventions	The counselling proceeded according to the framework of 5 A's :Assessment of								
	patients' prevailing PA habits and possibilities and willingness to increase PA, Advice								
	on sufficient PA regarding health and potential personal benefits of PA, Agreement on								
	PA goals and a weekly PA plan, Assistance at identifying PA barriers and suitable								
	exercise services and Arrangements for a control visit with a precise date including a								
	possibility to use a PA log for self monitoring. To support cooperation with other								
	health care and exercise workers, "Prex" could be used as a referral to								
	physiotherapists, nurses or exercise experts.								
Outcomes	Number of overall weekly PA sessions, Number of at least moderate-intensity weekly								
	PA sessions, Overall weekly PA in minutes, duration of at least moderate-intensity								
	weekly PA in minutes								
Notes									

Bias	Authors'	Support for judgement		
	judgement			
Random sequence	Unclear risk	The receptionists were not able to give the screening questionnaire to all the patients intended which may have		
bias)		caused selection bias		
Allocation concealment	Low risk	The physicians were randomised into the study groups		
(selection bias)				
Blinding of participants	Unclear risk	Single blinding as physicians were aware of the		
and personnel		intervention. Secondly, lack of time and staff holidays		
(performance bias)		prolonged the recruitment period, which may have		
		increased variation in PA due to seasonal reasons.		
Blinding of outcome	Unclear risk	PA data were based on self-reports, which		
assessment (detection		can be susceptible to over- or underreporting		
bias)				
Incomplete outcome	Low risk	Process evaluation was included in the analysis and the		
data (attrition bias)		dropout rate of patients at both 2- and 6-month follow-up		
		was small.		
Selective reporting	Low risk	All outcomes were reported		
(reporting bias)				

2. Atay, Torman & Yaman (2014) Methods The study was a randomised control trial with parallel groups Participants Primary care health centres from the metropolitan area of Antalya, Turkey, were chosen as study and intervention sites. Thirty nine primary health care centres were addressed and 33 centres were eligible for this study. Sedentary patients aged 50-70, without any physical health condition participated voluntarily to this study. Doctors from both groups attended the theoretical part (14 hours) of the exercise Interventions prescription course and the intervention group also attended the practical part of the course (10 hours). At the end of the 24 hours of training the doctors in the intervention group were given a "Doctor's Manual". In addition, the IG received instructional material and an equipment kit, which was prepared for the IG patients Outcomes The Seven Day Recall Physical Activity Questionnaire was used for estimating daily energy consumption. Physical performance measurements included the chair stand test, sit and reach test, back scratch test, balance test, and resistance band maximum repetition frequency test. In addition, heart rate, body weight, height, hip and waist circumference. measurements were also applied Notes

Risk of bias table

n'	A (1)	
Bias	Authors	Support for judgement
	judgement	
Random sequence	Low risk	Volunteer participants were evaluated at the primary health
generation (selection		care centres and recruited
bias)		
Allocation concealment	Low risk	Participating doctors in the study were divided into two
(selection bias)		groups, IG and CG, by random selection
Blinding of participants	Unclear risk	There was single blinding in the study. Doctors were not
and personnel		blinded to the intervention
(performance bias)		
Blinding of outcome	High risk	Turkish version of Borg's Scale (Rating of Perceived
assessment (detection		Exertion Scale-RPE) was given to the participants of the
bias)		study to enable them to measure the intensity of their
		activity. There was a possibility of assessment bias
Incomplete outcome	Unclear risk	There was explanation on withdrawal from the study.
data (attrition bias)		Unclear if the authors used an intention to treat analytic
		approach
Selective reporting	Low risk	All outcome results were reported
(reporting bias)		

3. Elley et al. (2003)

Methods	Cluster randomised controlled trial. Practices were randomised before systematic
	screening and recruitment of patients.
Participants	The study setting was 42 rural and urban general practices in one region of New
	Zealand. Subjects were all sedentary 40-79-year-old patients visiting their general
	practitioner during the study's recruitment period
Interventions	General practitioners were prompted by the patient to give oral and written advice on
	physical activity during usual consultations. Exercise specialists continued support by
	telephone and post. Control patients received usual care.

Outcomes	Change in physical activity, quality of life (as measured by the "short form 36" (SF-36)								
	questionnaire), cardiovascular risk (Framingham and D'Agostino equations), and								
	blood pressure over a 12-month period								
Notes									

Risk of bias table

Bias	Authors'	Support for judgement			
	judgement				
Random sequence generation	Low risk	Rolling recruitment of patients from each practice was			
(selection bias)		spread evenly from April 2000 to April 2001.			
Allocation concealment	Low risk	Participating general practices were stratified by size			
(selection bias)		and computer randomised them at a distant site			
		before recruiting patients			
Blinding of participants and	Low risk	Patients remained blind to whether they had been			
personnel (performance bias)		allocated to the intervention during screening for			
		activity and			
		enrolment.			
Blinding of outcome	Low risk	In measuring variables electronically, they used			
assessment (detection bias)		signed witness statements of verification to minimis			
		assessor bias.			
Incomplete outcome data	Low risk	No patients were excluded after enrolment. All			
(attrition bias)		outcome analyses were by intention to treat,			
		according to random allocation			
Selective reporting (reporting	Low risk	All variables assessed were reported regardless of			
bias)		significance			

4. Fortier et al. (2011)

Methods	The stuc	The study was a 25-week randomised controlled trial in Canada							
Participants	Patients were recruited from a large community-based primary care practice in								
	Ottawa,	Canada. Men ar	nd women aged 18 to 69 years who were visiting the practice						
	during	during the recruitment period for a scheduled nonemergency visit, who were							
	identifie	identified as inactive (<150 min of physical activity per week: yes-no) through							
	screenin	ig procedures ir	n the waiting room before their visit, and who reported no						
	uncontr	olled medical co	nditions						
Interventions	After receiving brief physical activity counselling from their provider, they were								
	randomised to receive 6 additional patient-centred counselling sessions over 3 months								
	from a physical activity counsellor (intensive-counselling group; n = 61), or no further								
	intervention (brief-counselling group; $n = 59$). The brief intervention consisted of the								
	first 4 A	first 4 As of the 7 As shared-care model for physical activity counselling in primary							
	care: (A	care: (A1) Addressing the subject of physical activity, (A2) Asking patients about their							
	physical	physical activity, (A3) Advising them to increase their physical activity while							
	personalizing the benefits, and (A4) Agreeing on a 1-month leisure-time physical								
	activity goal.								
Outcomes	Physical activity was measured using self-report and accelerometer.								
Notes									
Risk of bias table	able								
Bias		Authors'	Support for judgement						
	judgement								

Random sequence	Low risk	Between May and September 2005, trained research
generation (selection		assistants recruited patients from a large community-based
bias)		primary care practice in Ottawa, Canada
Allocation concealment	Low risk	The patients were randomly assigned to either the intensive-
(selection bias)		counselling group, which received an additional
		intervention from the physical activity counsellor, or the
		brief-counselling group, which received
		no further intervention.
Blinding of participants	High risk	Patient, research assistant, and physical activity counsellor
and personnel	C	blinding after randomizations was not feasible in this trial
(performance bias)		
Blinding of outcome	Low risk	Research assistant who conducted the physical and
assessment (detection		metabolic testing was blinded to group assignment
bias)		
Incomplete outcome	Low risk	In the intent-to-treat analyses was applied hence
data (attrition bias)		participants lost to follow up were also included in the
		analysis.
Selective reporting	High risk	Only data for patients with at least 4 completed days
(reporting bias)		(including at least 1 weekend day) for each time point
		were included in the statistical analysis

5. Goldstein et al. (1999)

Methods	The study design was a cluster randomised control trial. randomizations by practices							
	and not by patients. Twenty-four community-based primary care medical practices							
	were recruited into the study; 12 were randomised to the Intervention condition and							
	12	to	the	Control	conditio	on. Tł	ne intervention	
	perio	d was 4-7 v	veeks.					
Participants	The study was done in U.S. Potential subjects were contacted on the telephone to							
	deteri	mine eligik	oility, ob	tain informed c	onsent, and	gather base	eline information. We	
	exclue	ded patien	ts who v	vere too active (moderate ex	kercise for ->	>30 minutes at least 5	
	days	each week	or vigo	rous exercise for	r>20 minu	ites on at lea	ast 3 days per week),	
	were	not ambula	atory, and	d those unable to	o provide inf	formation or	n the telephone. Three	
	hund	red and fit	fty-five p	oatients were er	rolled in th	e study (18	1 in the Intervention	
	practi	practices and 174 in the Control practices). The mean age of the sample was 65.6 years,						
	a maj	a majority were women (65%),						
Interventions	а	patient-ca	antered	counselling	g appro	ach wl	hich emphasizes	
	interv	viewing	skills	that perm	nit tailori	ing of	the counselling	
	message. The counselling strategy utilizes the "5 As" (address the agenda, assess,							
	advise, assist, and arrange follow-up). The intervention was based on the							
	Transtheoretical Model which							
	considers the individual's performance of the desired behavior and the intention to							
	maintain or change this pattern of behavior							
Outcomes	Assessment was done At baseline, 6 weeks, and 8 months. Physical Activity Scale for							
	the Elderly (PASE):							
Notes								
Risk of bias table								
Bias		Authors'	S	upport for judg	ement			
		judgeme	nt					

Random sequence generation (selection	Low risk	Thirty-four physicians from 24 practices were recruited from Folio lists (20) of primary care practices (i.e. general internal		
Allocation concealment (selection bias)	Low risk	Practices were matched on whether they were solo or group practices; one practice in each pair of solo/group practices was randomised into the Intervention condition and the other into		
Blinding of participants and personnel (performance bias)	Low risk	the Control conditionThe study employed a double blinded approach. Office staff a all practices attended a half-hour administrative session. randomised to the Intervention condition, physicians attended a one-hour training session on physical activity counselling an provided physical activity counselling and activity counselling during a routine initial office visit and at a follow-u appointment scheduled within 4 weeks of the initia appointment. Physicians in the Control practices were not provided physical activity counselling training and were not expected to schedule patients for a follow-up visit for physica activity counselling. Patients were also blinded to the group they belonged to and had no knowledge of existence of the intervention		
Blinding of outcome assessment (detection bias)	High risk	Assessment was done by trained staff however there was no blinding of the outcome assessors		
Incomplete outcome data (attrition bias)	High risk	The final analysis excluded the patients that were lost to follow up and no explanation given to participants that did not complete the study. The study didn't employ an intention to treat analysis approach		
Selective reporting (reporting bias)	Low risk	all outcomes were reported in the analysis despite the results being significant or not.		

6. Grandes et al. (2009)

Methods	The study was a pragmatic, cluster randomised controlled clinical trial in Spain from							
	October 2003 to December 2004. Fifty-six Spanish family physicians were randomised							
	to either the intervention(n=29) or standard care (n=27) arm of the trial. The study took							
	a duration of six months							
Participants	The study was done at 11 public primary care centres with family physicians as							
	allocation units. All 15 research groups of the Spanish Preventive Services and Health							
	Promotion Primary Care Research Network were invited to participate. After signing							
	a collaboration consent form, physicians were randomised to either the PEPAF or							
	usual care (control) arm of the trial in a 1:1 ratio using computer-generated random							
	numbers stratified by centre and provided by a central site. Family physicians							
	recruited patients aged 20 to 80 years, who did not meet the recommended aerobic							
	physical activity levels (moderate-intensity physical activity for30 minutes 5 d/wk. or							
	vigorous intensity activity for 20 minutes 3 d/wk.).							
Interventions	Physicians allocated to the PEPAF arm provided brief advice							
	and educational materials to all patients and offered an additional 15-minute							
	appointment to prescribe an individualized							
	physical activity plan. Control group physicians delivered standard care and delayed							
	any new systematic intervention related to physical activity until the end of the study							

Outcomes	The	primary	outcome	was	7-Day	Physical	Activity
	Recall (F	AR) measured	d as moderate	and	vigorous activi	ty (minutes	/week) and
	moderate	e and vigorous	activity (MET	-h/we	ek). Secondary	outcome mea	asures were
	maximur	n oxygen uptal	ke (V [·] O2max)				

Notes Risk of bias table

Bias	Authors'	Support for judgement
	judgement	11 , 5
Randomsequencegeneration(selectionbias)	Low risk	After meeting requirements for inclusion, 4317 patients were randomly included in the study. At baseline, both groups had similar values of outcome variables.
Allocation concealment (selection bias)	Low risk	physicians were randomised to either the PEPAF or usual care (control) arm of the trial in a 1:1 ratio using computer- generated random numbers stratified by centre and provided by a central site.
Blinding of participants and personnel (performance bias)	Unclear risk	Single blinded study as the assessors were not blinded to the intervention
Blinding of outcome assessment (detection bias)	Low risk	physicians assessed the patients' physical activity with assistance of a computerized algorithm. To avoid recruitment bias, candidates to be assessed by their physicians were systematically sampled by research nurses from the list of patients scheduled for consultation.
Incomplete outcome data (attrition bias)	Low risk	The study employed intention to treat analysis. The 6-month follow-up visit was completed by 81% of patients. Patients lost to follow-up did not differ between the groups, except by age and social class
Selective reporting (reporting bias)	Low risk	All outcomes assessed were reported regardless of significance of results. For example, there were no significant differences in secondary outcomes between groups

7. Grandes et al. (2011)

Methods	A 24 mo	nths pragmatic,	cluster randomised clinical trial initiated in October 2003	
Participants	The study was conducted in eleven public primary care centres in Spain. Fifty-six			
	general	general practitioners (GPs) were randomly assigned to intervention (29) or standard		
	care (27)	groups. They a	ssessed the physical activity level of a systematic sample of	
	patients	in routine practi	ce and recruited 4317 individuals (2248 intervention and 2069	
	control)	who did not me	et minimum physical activity recommendations	
Interventions	Intervention GPs provided advice to all patients and a physical activity prescription to			
	the subgroup attending an additional appointment (30%). A third of these			
	prescriptions were opportunistically repeated. Control GPs provided standard care			
Outcomes	Primary outcome measure was the change in self-reported physical activity from			
	baseline to six, 12 and 24 months. Secondary outcomes included cardiorespiratory			
	fitness and health-related quality of life.			
Notes				
Risk of bias table				
Bias		Authors'	Support for judgement	
		judgement		

Random sequence	Low risk	General practitioners were invited to participate through
generation (selection		the Spanish Preventive Services and Health Promotion
bias)		Primary Care Research Network
Allocation concealment	Low risk	a total of 70 family physicians as allocation units from 13
(selection bias)		health centres were randomised before patient recruitment,
		in a 1:1 ratio using computer-generated random numbers
		provided by a central site
Blinding of participants	Unclear risk	The study was single blinded as general practitioners were
and personnel		aware of the intervention but the patients were not
(performance bias)		knowledgeable on the intervention being administered
Blinding of outcome	Low risk	Trained nurses working in exercise laboratories who
assessment (detection		performed baseline and follow-up measurements at 6, 12
bias)		and 24 months were blinded to the allocation group of the
		participants
Incomplete outcome	Low risk	A telephone recall system was used to improve follow-up
data (attrition bias)		rates
Selective reporting	Low risk	All outcomes were reported regardless of significance of
(reporting bias)		results

8. Green et al. (2002)

Methods	This was a randomised controlled trial, conducted from 1997 to 1998, in Washington
	State U.S.
Participants	The setting was one family physician's patients in a suburban community. patients
	aged 18 to 65 who were recruited from a mailed health risk assessment. Patients were
	eligible for this study if they were inactive (exercised 15 minutes per day, even if they
	exercised daily) and specified that they were interested in increasing their exercise in
	the next 6 months
Interventions	Three sessions of telephone-delivered motivational counselling
Outcomes	Physical activity score (11-item Physician-Based Assessment and counselling for
	Exercise [PACE]) 6 months after the intervention
Notes	

Bias	Authors'	Support for judgement
	judgement	
Random sequence	Unclear risk	Patients were randomly recruited from 1330 adults
generation (selection bias)		
Allocation concealment	Low risk	A total of 316 patients were included in this study and
(selection bias)		were randomly assigned using a standard random
		number generator to be offered the intervention or be a
		control.
Blinding of participants	Unclear risk	This study was single blinded because patients were
and personnel		blinded to group allocation, but health providers were not
(performance bias)		blinded.
Blinding of outcome	Low risk	All controls and all those in the intervention group (both
assessment (detection		those who received the intervention and those who did
bias)		not) were questioned at baseline and 6 months
		after enrolment in the intervention by a study interviewer,
		independent of the health educator, who was blinded as to
		the participants' intervention status.

Incomplete outcome data Unclear risk		Unclear risk	Treatment groups were analysed as randomised in an
(attrition bias)			intent-to treat analysis however In secondary analysis,
			differences between treatment groups were analysed as
			treatment received.
Selective	reporting	Unclear risk	All outcome results were reported regardless of
(reporting bias)			significance.

9. Hillsdon et al. (2002)

Methods	The study was a randomised control trial in Wellingborough, England between		
	February 1996 and May 1997		
Participants	Participants were included in the study if they were aged 45-64 years and had less than		
	four occasions of moderate intensity physical activity.1658 middle aged men and		
	women were randomly assigned to 30 minutes of brief negotiation or direct advice in		
	primary care or a no-intervention control group.		
Interventions	The interventions include direct advice group and brief negotiation. All intervention		
	subjects were telephoned at set intervals following health check. The direct advice		
	group received more advice about the importance of a physically active lifestyle		
Outcomes	The primary outcome was self reported physical activity at 12 months. The secondary		
	outcomes were blood pressure and body mass index		
Notes			
D. 1 (1.1) 1.1			

Risk of bias table

Bias	Authors'	Support for judgement
	judgement	
Random sequence generation	Low risk	It was random recruitment
(selection bias)		
Allocation concealment	Unclear risk	Subjects were randomised by one of the authors to
(selection bias)		one of the three arms
Blinding of participants and	Unclear risk	The study did not mention if the control groups
personnel (performance bias)		were blinded to the intervention.
Blinding of outcome assessment	Unclear risk	The primary outcome was reported using self
(detection bias)		report. Bias is unclear
Incomplete outcome data	Low risk	Missing blood pressure and weight and height
(attrition bias)		measures at follow up were assigned the baseline
		values.
Selective reporting (reporting	Low risk	All outcomes were reported
bias)		

10. Kerse et al. (2005)

Methods	The study was a cluster randomised, controlled trial that was done over a 12months
	period
Participants	The study involved one hundred seventeen doctors in 42 primary care practices in the
	Waikato region of New Zealand. Two hundred seventy sedentary primary healthcare
	patients aged 65 and older participated. To be included, patients had to be
	participating in less than 30 minutes of at least moderate intensity physical activity on
	5 or more days per week, be aged 40 to 80, and intend to stay in the region for at least
	12 months. Patients were excluded if they were unable to comprehend the informed
	consent or were suffering from an unstable cardiovascular, debilitating, or progressive
	illness

Interventions	Patients in intervention practices prompted their primary care doctors or practice
	nurse to deliver brief activity counselling. A "Green Prescription" was written
	involving the negotiation of activity goals. Trained exercise specialists from a regional
	sports foundation gave follow-up telephone support over 3 months.
Outcomes	Leisure moderate and vigorous physical activity, total energy expenditure, systolic
	and diastolic blood pressure

Notes Risk of bias table

Bias	Authors'	Support for judgement
	judgement	
Random sequence	Low risk	All primary care doctors in the Waikato region were invited
generation (selection		to participate hence randomization was guaranteed.
bias)		
Allocation concealment	High risk	Patients and doctors could
(selection bias)		not be blinded to group allocation.
Blinding of participants	Unclear risk	Single blinded study as only participants were blinded to
and personnel		the intervention. The doctors in the control group were
(performance bias)		aware of the intervention however, received no prompt to
		deliver the Green Prescription
Blinding of outcome	Low risk	Outcomes were ascertained by self-completion
assessment (detection		questionnaires at baseline and follow-up.
bias)		
Incomplete outcome	Low risk	All outcomes were assessed using intention-to-treat
data (attrition bias)		analyses. Participants unable to follow up were assigned
		their baseline status on follow up, because this was
		considered the most conservative estimate
Selective reporting	Unclear risk	All outcomes were reported including those with non-
(reporting bias)		significant results e.g. blood pressure

11. Kolt et al. (2007)

Methods	The study was a randomised, controlled trial conducted over a 12-month period.			
Participants	The study setting was three primary care practices from different socioeconomic			
	regions of Auckland, New Zealand. One hundred and eighty-six low-active adults			
	(aged 65) recruited from their primary care physicians' patient databases.			
Interventions	The intervention was Eight telephone counselling sessions			
	over 12 weeks based on increasing physical activity. Control patients received usual			
	care.			
Outcomes	Change in physical activity (as measured using the Auckland Heart Study Physical			
	Activity Questionnaire)			
Notes				
Diel, of hiss table				

Risk of bias table Bias Authors' Support for judgement judgement Those patients for whom physical activity was not Random sequence Low risk contraindicated and were contactable at the address and generation (selection bias) telephone number on the practice database were invited to participate in the study via a letter from their primary care physician and follow-up telephone call from the practice where necessary.

Allocation concealment	Low risk	After baseline assessment, one of the researcher's computer
(selection bias)		randomised (simple randomisation) participants to control
		or intervention groups. Another researcher (NG) generated
		the allocation sequence.
Blinding of participants	Unclear risk	Unclear if the participants were blinded of the intervention
and personnel		but the physicians were aware of the intervention. Hence it
(performance bias)		was single blinded
Blinding of outcome	Low risk	The assessor was blinded to group allocation
assessment (detection		
bias)		
Incomplete outcome	Unclear risk	Unclear how data from patients lost to follow up was
data (attrition bias)		handled.
Selective reporting	Low risk	All outcome data was reported
(reporting bias)		

12. Lawton et al. (2008)

Methods	The study was a randomised controlled trial.
Participants	The setting was in 17 primary care practices in Wellington, New Zealand. Eligibility
	criteria included women 40-74 who were physically inactive
Interventions	Brief physical activity intervention led by nurse with six-month follow-up visit and
	monthly telephone support over nine months.
Outcomes	Primary outcome measure was the proportion of those achieving the recommended
	150 minutes of at least moderate intensity physical activity, as assessed by the long
	form of the physical activity questionnaire. Secondary outcomes included quality of
	life assessed with the short form 36 questionnaire (SF-36); weight, waist circumference,
	and blood pressure
Notes	

Bias	Authors'	Support for judgement
	judgement	11 , 0
Random sequence	Low risk	A researcher not involved in the recruitment process
generation (selection bias)		carried out computer generated block randomisation
Allocation concealment	Low risk	After baseline measures the nurse opened
(selection bias)		sequentially numbered opaque envelopes containing the
		allocated treatment group (intervention or control).
Blinding of participants	High risk	There was no blinding of the general practitioner or the
and personnel	_	participants to the intervention.
(performance bias)		
Blinding of outcome	Low risk	Nurses assessing participants at 12- and 24-month follow-
assessment (detection		up visits were blind to group allocation, and participants
bias)		were asked not to discuss group allocation with the
		assessing nurse
Incomplete outcome data	Low risk	They carried out an intention to treat analysis of all
(attrition bias)		participants enrolled in the study according to allocation of
		randomisation
Selective reporting	Unclear risk	All outcomes were reported
(reporting bias)		-

13. Lewis & Lynch (1993) Methods The study was a randomised control trial conducted from march through June 1991 in the US The study was conducted in the ambulatory practice of the department of Family **Participants** medicine, the University of Colorado Health Science Centre. Twenty-four family medicine residents were randomly assigned to the experimental or control groups. Subjects included outpatients age 18 and older who were scheduled to see a resident. A sample of 396 patients were admitted to the study. The intervention included 2 to 3 minutes of exercise advice, the distribution of an Interventions educational handout and the promise of a 1 month follow up phone call from a staff person. The exercise advice consisted of three steps of interaction with the patient, ASK about exercise, ASSESS the response and ADVISE accordingly. Outcomes physical activity minutes/session, PA Times/week, PA minutes/week Notes

Risk of bias table

Bias	Authors'	Support for judgement
	judgement	11 , 0
Random sequence generation	Low risk	Selection of participants was in a random
(selection bias)		sequence.
Allocation concealment (selection	Low risk	family medicine residents were randomly
bias)		assigned to the experimental or control groups
Blinding of participants and	Low risk	The participants and family medicine residents
personnel (performance bias)		in the control were blinded to the intervention
Blinding of outcome assessment	Unclear risk	The assessment was self reported. There risk of
(detection bias)		bias is unclear
Incomplete outcome data	Unclear risk	Unclear how attrition was handled
(attrition bias)		
Selective reporting (reporting	Low risk	All outcome data was reported
bias)		

14. Little et al. (2004)

Methods	The study design was randomised controlled (2 X 2 X 2) factorial trial. The study was
	approved by the Southampton and Salisbury Ethics Committees. Patients were
	randomly chosen from practice databases and were randomly assigned to one of eight
	groups. The study took 3 years from 1999 to 2002
Participants	The study was conducted in Britain in four practice settings. Patients were included if
	they had one or more risk factors for coronary heart disease: a diagnosis by a GP of
	hypertension or hyperlipidaemia, a body mass index >25, or diabetes. Patients were
	excluded if they had coronary heart disease, if they were unable to perform moderate
	exercise (for example, if they had severe left ventricular failure), if they were unable to
	complete the questionnaire (for example, because of dementia). Participants who were
	both male and female and above the age of 18 years were included in the study.
Interventions	Patients could be assigned to no intervention, a single intervention, or any combination
	of interventions. The following interventions were used in the study. Exercise
	prescription-GPs briefly discussed the benefits of exercise, targets, how to start, and
	anticipating relapse, and wrote a prescription for 30 minutes, 5 times a week, of brisk
	walking (or equivalent). Counselling session- Nurses discussed the same issues as
	with exercise prescription. They also had a detailed motivational discussion (based on
	the theory of planned behaviour, which addresses attitudes and perceived behavioural

	control), identifying a precise time and place to start ('behavioural rehearsal'), and		
	agreed and signed a contract. Booklet- The Health Education Authority booklet		
	Getting active, feeling fit was used.		
Outcomes	Assessments were carried out at baseline and after 1 month. The following data were		
	collected: height; weight (using digital scales); blood pressure, using a semi-automated		
	validated Omron oscillometer sphygmomanometer; cholesterol level, i.e. non-fasting		
	total cholesterol, high density lipoprotein (HDL), and cholesterol/HDL ratio; the well-		
	validated Godin questionnaire, which multiplies the number of		
	episodes of exercise by relative energy expenditure in each; 'stage of change'; physical		
	fitness/performance. The two measures of exercise performance used to assess fitness		
	were: the Canadian Home Step Test, which took about 20 minutes to explain and		
	perform; and the 6-minute walking test, in which participants are asked to walk or run		
	as fast as they can around a circuit of known distance (200 meters on the pavement		
	around the surgery) for 6 minutes, and the distance walked predicts health outcomes		
Mataa			

Notes Risk of bias table

Bias	Authors'	Support for judgement
	judgement	
Random sequence generation (selection bias)	Low risk	Patients were randomly chosen from practice databases if they had one or more risk factors for coronary heart disease
Allocation concealment (selection bias)	Low risk	Patients were randomly assigned to one of eight groups, which were defined by three intervention factors, in a balanced 2 X 2 X 2 factorial design, by opening a sealed opaque numbered envelope that had been prepared previously at the trial centre by the research nurse.
Blinding of participants and personnel (performance bias)	Unclear risk	The study has not clearly explained if the environment in which the participants was different, however it has tried to exclude participants with covariates that would possibly affect their fair participation.
Blinding of outcome assessment (detection bias)	Unclear risk	The assessors did not take part in the intervention, and made assessments without reference to the intervention group. Full blinding of assessors was not possible, given that this was an open trial, but patients were asked not to say what they had done
Incomplete outcome data (attrition bias)	Low risk	Missing values at follow-up were replaced with baseline measurements for intention-to-treat analyses, making the most conservative assumption that they would not have changed
Selective reporting (reporting bias)	Low risk	The study reported results despite being insignificant. For example there was relative increase in the cholesterol/HDL ratio in the prescription group compared to the control cholesterol/HDL ratio (0.34) in the control group, which may therefore be a chance finding, since the control group exercised less than the other groups

15. Mutrie et al. (2012)

Methods	Two-arm (intervention/control) 12-week randomised controlled trial with a 12-week
	follow-up for the intervention group.

Participants	The setting was in one general practice in Glasgow, UK. Participants were aged ≥ 65
	years. Inclusion criteria were living independently and not meeting current PA
	recommendations.
Interventions	The intervention group received two 30-minute physical activity consultations from a
	trained practice nurse, a pedometer and a walking programme. The control group
	continued as normal for 12 weeks and then received the intervention. Both groups
	were followed up at 12 and 24 weeks.
Outcomes	Step counts were measured by sealed pedometers and an activPAL TM monitor.
	Psychosocial variables were assessed and focus groups conducted.
Notes	

Risk of bias table

Bias	Authors'	Support for judgement
	judgement	
Random sequence	Low risk	Random patients screen by GP and meet the inclusion
generation (selection		criteria
bias)		
Allocation concealment	Low risk	Randomization was performed
(selection bias)		using an ordered set of sealed envelopes containing group
		allocations. Allocations were made by an independent
		member of the research team and inserted into envelopes in
		a random order
Blinding of participants	Low risk	The nurse and a research assistant performing the
and personnel		randomisation were not blinded to group allocation but all
(performance bias)		other researchers were.
Blinding of outcome	Low risk	The pedometer and activPAL were fitted
assessment (detection		by the practice nurse who had been appropriately trained
bias)		by the research team.
Incomplete outcome data	Unclear risk	Data processing errors resulted in loss of activPAL data for
(attrition bias)		four participants (one intervention, three controls) at
		baseline and for one participant (intervention) at week 24.
Selective reporting	Low risk	All outcome results were reported
(reporting bias)		

16. Norris et al. (2000)

Mathada	This study is a randomized controlled trial Dravidars were randomized to the control
Methous	This study is a randomised controlled trial. Providers were randomised to the control
	or the intervention groups. All patients seeing a given provider were randomised into
	groups. The study took 6 months.
Participants	The study setting was Group Health Cooperative of Puget Sound, a staff model health
	maintenance organisation in Pacific Northwest. Study patients were recruited from
	consecutive patients, male and female of age 30 years or older, who called their
	primary care physician to schedule a well visit. Patients were excluded from the study
	if they had significant cognitive impairments, did not speak English, were pregnant or
	had cardiovascular, respiratory or metabolic disease
Interventions	Intervention physicians were trained to deliver exercise counselling protocol at the
	index visit, and one reminder telephone that occurred at 1 month.
Outcomes	The outcomes included Physical activity index (Kcal/week), Leisure score (0-130), Total
	physical activity (minutes/week), Total minutes walking per week
Notes	

Bias	Authors'	Support for judgement
	judgement	
Random sequence	Low risk	Patients were recruited from consecutive age 30-year old's
generation (selection		who called their primary care physician to schedule a well
bias)		visit (a complete physical or female gynaecological
		investigation)
Allocation concealment	Low risk	Providers were randomised to the control or the intervention
(selection bias)		groups. The patient control and intervention groups were
		similar at baseline for demographic characteristics and
		physical activity levels.
Blinding of participants	Unclear risk	There was single blinding of the patients only, the physicians
and personnel		were not blinded on the intervention and were not monitored
(performance bias)		
Blinding of outcome	Unclear risk	More part-time physicians were in the intervention group
assessment (detection		than the control group
bias)		
Incomplete outcome	High risk	The authors analysed the data using the per protocol analysis
data (attrition bias)		approach and ignored data for patients who had been lost to
		follow up
Selective reporting	High risk	The authors selectively reported data. When they realised
(reporting bias)		that there were no significant differences in any of the
		measures of physical activity, they combined two
		interventions groups and reported the results verses a control
		comparison.

17. Pears et al. (2016)

Methods	The study was a randomised controlled trial in which interventions were delivered on
	a randomised weekly basis between May 2013 and February 2014
Participants	Participants were recruited from eight NHS primary care practices in urban and rural
	areas in the East of England. Participants were eligible for the trial if they were eligible
	for the Health Check: i.e. aged 40–74 years and not previously diagnosed with heart
	disease, stroke, diabetes, or kidney disease. Participants were excluded if they had no
	working knowledge of English
Interventions	Participants were randomised to a Motivational $(n = 83)$, Pedometer $(n = 74)$, or
	Combined (n = 80) intervention, delivered immediately after a preventative health
	check in primary care, or control (Health Check only; n = 157).
Outcomes	physical activity, measured by accelerometer at 4 weeks.
Notes	

Bias	Authors'	Support for judgement
	judgement	
Random sequence generation (selection bias)	Low risk	The randomisation list was constructed by programming a simple block randomisation routine in R software
Allocation concealment (selection bias)	Low risk	Within each general practice, each calendar week of Health Checks was randomised to one of the four trial arms, such that all participants scheduled to receive a Health Check consultation in the same week were allocated to the same trial arm

Blinding of participants	Low risk	Participants were blind to allocation at the time of booking
and personnel		their appointment, as were the practice staff who arranged the
(performance bias)		appointment
Blinding of outcome	High risk	The general practitioners were aware of the intervention and
assessment (detection		hence possible bias
bias)		
Incomplete outcome	High risk	Follow up data was analysed based on availability. hence the
data (attrition bias)		study did not apply an intention to treat analysis approach
Selective reporting	Low risk	All outcome data was reported
(reporting bias)		

18. Petrella et al. (2003)

Methods	randomised controlled trial; baseline assessment and intervention delivery with
	postintervention follow-up at 3, 6, and 12 months
Participants	Four large (5000 active patient files) academic, primary care practices: three in urban
	settings and one in a rural setting, each with four primary care physicians; two clinics
	provided the Step Test Exercise Prescription (STEP) intervention and two provided
	usual care control. A total of 284 healthy community-dwelling patients (72 per clinic)
	aged 65 years were recruited in 1998–1999 in Canada. Participants' mean age was 73
	years
Interventions	Physicians in the intervention group were given published exercise counselling
	guidelines (exercise prescription instrument), a paper describing the benefits of
	exercise, guidelines for delivery and training in interpretation of the step test data to
	determine patient aerobic capacity (VO2max), including the prescription of an exercise
	training heart rate. Physicians in the control group were instructed to provide subjects
	with exercise counselling and prescription per their "usual care," with the addition of
	the ACSM guidelines and the benefits of exercise
Outcomes	The primary outcome measure was aerobic fitness (VO2max). Secondary outcomes
	included predicted VO2max from the STEP test, exercise self-efficacy (ESE), and
	clinical anthropometric parameters.
Notes	

Bias	Authors'	Support for judgement
	judgement	
Random sequence generation (selection bias)	Low risk	Patients were identified in two ways over 6 months. First, over a 2-month recruitment period, clinic staff identified potentially eligible patients opportunistically from the regular daily register. Second, a clinic-produced list of patients meeting the eligibility criteria was utilized until 72 patients from each clinic were identified. All staff were blinded during recruitment.
Allocation concealment (selection bias)	Low risk	Subjects were then randomised to either STEP or control by a computer program, and scheduled to meet with a clinic family physician corresponding to their group assignment for exercise counselling
Blindingofparticipantsandpersonnel(performance bias)	High risk	There was patient contamination because both STEP and control groups were given a list of available facilities for physical activity participation in their community. At all times, patients

		in both groups were free to choose where and how they would	
		exercise	
Blinding of outcome	High risk	There was no blinding of outcome assessment because the same	
assessment (detection	-	physicians who delivered initial counselling saw the patients at	
bias)		3, 6, and 12 months, there was a possibility of positive	
		assessment bias	
Incomplete outcome	High risk	The study excluded results from participants lost to follow up,	
data (attrition bias)		therefore did not employ intention to treat strategies	
Selective reporting	Low risk	All outcome results were reported even if they yielded non-	
(reporting bias)		significant results	

19. Petrella, Lattanzio, Shapiro & Overend (2010)

Mathada	study was a 12 month dustar randomized trial
Methods	study was a 12-month cluster randomised that
Participants	The setting was forty family practices in 4 regions of Canada. Healthy, community-
	dwelling men (48%) and women (52%) with a mean (SD) age of 64.9 (7.1) years (range
	55 to 85 years). There were a total of 193 participants in the intervention group and 167
	in the control group.
Interventions	Intervention physicians were trained to deliver a tailored exercise prescription and a
	trans-theoretical behaviour change counselling program. Control physicians were
	trained to deliver the exercise prescription alone.
Outcomes	Predicted cardiorespiratory fitness, measured by predicted maximal oxygen
	consumption (pVO2max), and energy expenditure, measured by 7-day physical
	activity recall.
Notes	

Risk of bias table

Bias	Authors'	Support for judgement
	judgement	
Random sequence	Low risk	Random selection of family physicians willing to
generation (selection bias)		participate
Allocation concealment	Low risk	Family physicians agreeing to participate were
(selection bias)		randomised to either the intervention or the control
		group.
Blinding of participants and	Unclear risk	Single blinding because only patients were blinded to
personnel (performance bias)		the intervention, however family physicians were fully
		aware of the intervention
Blinding of outcome	Low risk	Assessment were collected by trained staff members;
assessment (detection bias)		then step tests, supervised by the family physicians.
Incomplete outcome data	Low risk	Dropouts were not replaced, as we performed an
(attrition bias)		intent-to-treat analysis
Selective reporting (reporting	Unclear risk	All outcomes were reported
bias)		

20. Pfeiffer et al. (2001)

Methods	The study was a randomised control trial. Data were collected during a 28-week period		
	from late February through August 1999.		
Participants	During their regular office visits to the geriatrician, adults aged 60 years and older		
	were informed of the study and invited to participate. It was conducted in a geriatric's		
	ambulatory clinic, which is contiguous to a medical school in rural Appalachia, Ohio.		

	Exercise baseline data were collected on 76 individuals; 49 (44 women and 5 men) of
	these individuals were enrolled in the study
Interventions	Enrolees were randomly assigned to either the green prescription group (n = 24) or the
	verbal advice only group (n =25) using a table of random numbers. The intervention
	involved exercise advice. This exercise advice was given verbally to all participants by
	the physicians. Then, the physician opened an envelope that indicated if that patient
	was in the group to receive further written exercise advice. Those patients placed in
	this group had their goals written on a green prescription form.
Outcomes	The outcomes were leisure physical activity in minutes/week and total physical
	activity in minutes /week

Notes Risk of bias table

Bias	Authors'	Support for judgement
	judgement	
Random sequence	Unclear risk	Unclear if there were participants were selected based
generation (selection bias)		on a random sequence
Allocation concealment	Low risk	Enrolees were randomly assigned to either the green
(selection bias)		prescription group or the verbal advice only group using
		a table of random numbers.
Blinding of participants and	Low risk	It was single blinded as the physicians were aware of the
personnel (performance		intervention, however the patients were not aware
bias)		
Blinding of outcome	Unclear risk	The study was self reported. The possibility of bias in
assessment (detection bias)		assessment of outcome is unclear
Incomplete outcome data	Low risk	The analysis was also performed on an intention-to-treat
(attrition bias)		basis assuming no change in exercise status for the two
		participants who withdrew from the study
Selective reporting	Low risk	All outcomes were reported regardless of significant or
(reporting bias)		non significant results.

21. Reid & Morgan (1979)

Methods	The study was a randomised control trial that was done within a six months period in		
	Canada		
Participants	One hundre	d and twenty-for	ur firefighters, age 24 to 56, volunteered for this study.
	Through me	dical screening, a	uthors evaluated the safety and suitability of participation
	for each of t	he volunteers. Fi	fteen per cent of those who came to the screening were
	ineligible for	the study; those	e with contraindications for strenuous exercise and those
	already in pł	nysical activity pr	ograms were excluded from the study.
Interventions	The intervention included a printed exercise instruction and a ten-minute consultation		
	with a physician, a one-hour period of film and discussion and knowledge of pulse		
	taking, quantifying, recording of daily exercise, and reporting of this information.		
Outcomes	Maximum oxygen uptake (VO2) was estimated on three occasions (start, three months,		
	six months) using a bicycle ergometer, a stepwise increase of workload, and the Astrand		
	nomogram		
Notes			
Risk of bias table	e		
Bias		Authors'	Support for judgement
		judgement	

Random sequence	Low risk	Participants randomly volunteered for this study.
generation (selection bias)		
Allocation concealment	Low risk	Participants were randomly allocated to a control and
(selection bias)		two treatment groups.
Blinding of participants and	Low risk	Participants were randomly and received no further
personnel (performance		instructions on the intervention
bias)		
Blinding of outcome	Unclear risk	The participants self monitored, but the participants
assessment (detection bias)		were blinded. Hence it was a single blinded study as
		physicians were not blinded to the intervention.
Incomplete outcome data	Low risk	Missing cases were assigned a zero per cent change in
(attrition bias)		their aerobic condition and are included in the analysis.
Selective reporting	Low risk	All outcomes were reported
(reporting bias)		-

22. Swinburn et al. (1998)

Methods	The trial involved a randomised, controlled design conducted in a period of over a 13
	weeks period
Participants	The study was carried out in two New Zealand urban centres (Auckland and
	Dunedin). Thirty-seven general practitioners underwent a training session on
	assessing and prescribing physical activity. They recruited patients who, in their
	judgment, were likely to benefit from an increase in physical activity and were able to
	increase their exercise over the following 6 weeks. Participants' mean age was 49 years
	(SD = 15). Participants (n = 491) were randomised to green prescription (n = 239) or
	verbal advice only (n = 252),
Interventions	Baseline data on exercise levels were collected by general practitioners using a
	standard questionnaire. For each participant, goals to increase physical activity
	(mainly centred around walking) were established. After the verbal advice had been
	given, the general practitioner opened an envelope that randomised the participant
	(within general practitioners) to having the goals written down or not. After 6 weeks,
	follow-up telephone interviews were conducted by trained interviewers using the
	same set of questions.
Outcomes	Physical activity duration in minutes/week
Notes	

Bias	Authors'	Support for judgement
	judgement	
Random sequence generation	High risk	General practitioners recruited patients who, in their
(selection bias)		judgment, were likely to benefit from an increase in
		physical activity
Allocation concealment	Low risk	general practitioners randomised the participant into
(selection bias)		groups
Blinding of participants and	Low risk	Participants were unaware of the intervention but
personnel (performance bias)		general practitioners were not blinded
Blinding of outcome	Unclear risk	Interviewers were unaware of the randomisation
assessment (detection bias)		group of participants
Incomplete outcome data	Unclear risk	Analysis was also performed on an intention-to-treat
(attrition bias)		basis assuming no change in exercise status among
		those lost to follow-up.

Selective reporting (reporting	Unclear risk	All outcome data was available
bias)		

23. Sørensen et al. (2008)

Methods	The study was conducted as a randomised trial in 2005–2006
Participants	All the patients referred to the Exercise on Prescription scheme by their GP in the
	counties of Vejle and Ribe, Denmark, were eligible for the trial
Interventions	The high-intensive exercise on prescription group received 4 months of group-based
	supervised training and attended five motivational counselling sessions. The low-
	intensive group only attended four motivational counselling sessions.
Outcomes	Maximal oxygen uptake (VO2max), Body mass index (BMI), Physical activity
	(METh/day)
Notes	

Risk of bias table

Bias	Authors'	Support for judgement
	judgement	11 , 0
Random sequence	Low risk	All patients referred to the Exercise on prescription scheme
generation (selection		were eligible for the study and were offered participation in
bias)		the randomised study during the baseline health profile by
		the physiotherapists
Allocation concealment	Low risk	Randomization was carried out by the first author by means
(selection bias)		of concealed envelopes containing the name of the group.
Blinding of participants	Unclear risk	The participants were blinded to the intervention, but it was
and personnel		impossible to blind the healthcare providers
(performance bias)		
Blinding of outcome	High risk	The physiotherapists carried out the physiological measures
assessment (detection	_	and handed out questionnaires for the self-reported
bias)		measures
Incomplete outcome	Low risk	missing data were replaced with the participant's mean
data (attrition bias)		value from the other questions
Selective reporting	Unclear risk	All outcome results were reported
(reporting bias)		

24. Tylor & Fox (2005)

Methods	The study was a randomised control trial. Randomization, on the basis of a random-
	numbers table, took place at the end of the first assessment, with a 7:3 greater
	likelihood of participants being referred into the exercise program than into the control
	group to offset anticipated non adherence to the exercise program
Participants	142 patients were randomised women, ages 40-70 years, with one or more of three
	coronary heart disease (CHD) risk factors (i.e., being a smoker, hypertensive, or
	overweight) were identified from primary care medical records. There was no gender
	or age bias in U.K
Interventions	Patients were asked to visit the local leisure centre (exercise facility) and arrange an
	appointment for an introductory session to start a 10-week program with 2 sessions
	per week. At the end of the 10 weeks, a progress report was returned to the
	participant's GP, and participants were encouraged to maintain a physically active
	lifestyle
Outcomes	The outcome measures included total energy expenditure, moderate PA in minutes,
	Secondary outcomes included Body weight, BMI, fat, and heart rate

Notes		
Risk of bias table		
Bias	Authors'	Support for judgement
	judgement	
Random sequence	Low risk	Patients were randomly selected
generation (selection		
bias)		
Allocation	Low risk	Randomization was done on the basis of random numbers into
concealment		the intervention and control groups
(selection bias)		
Blinding of	Unclear risk	Participants were blinded to the intervention; however, the
participants and		physicians were not blinded.
personnel		
(performance bias)		
Blinding of outcome	Unclear risk	Unclear who was assessing outcomes, hence unable to
assessment (detection		determine performance bias
bias)		
Incomplete outcome	Low risk	For participants with missing data at the 16- and 37-week
data (attrition bias)		assessments, values were imputed from the previous
		assessment with complete data to perform an intent-to-treat
		analysis.
Selective reporting	Unclear risk	Data on type, duration, and intensity of exercise in each session
(reporting bias)		within the cardiovascular exercise room were self-reported on
		a card at each visit by participants within the scheme. Further
		scrutiny revealed that these data were often too subjective and
		incomplete to use in any data analysis.

b. Characteristics of excluded studies

Activity (2001)

T

Reason for	The study did not have a control group receiving usual care or placebo
exclusion	

Armit et al. (2009)

Reason for	The study was excluded because it did not have a control group that received
exclusion	either a placebo or no intervention

Bennet et al. (2008)

Reason for	The study was excluded because physical activity counselling intervention was
exclusion	administered by a counsellor and not a health provider

Bucholz & Purath (2007)

Reason for	This descriptive exploratory study examined factors related to physical activity
exclusion	counselling practices

Bull & Jamrozik (1999) Reason for exclusion The study was excluded due to inadequate data for use in the meta analysis

Burn, Camaione & Chartterton (2000)

Reason for	The study was excluded because it was a survey
exclusion	

Calfas et al. (1996)

Reason for	Intervention students attended 15 weekly 50-minute lectures led by 1 behavioral
exclusion	and 1 exercise science faculty member, hence was excluded because the
	intervention was not delivered by a healthcare provider.

Cunningham et al. (1987)

Reason for	The intervention was not primary care based
exclusion	

Dubbert et al. (2002)

Reason for	The study was excluded because baseline data was missing on the variables of
exclusion	interest

Duru et al. (2010)

Reason for	The study used a faith-Based Physical Activity Intervention
exclusion	

Fortier et al. (2006)

Reason for	The study is missing results section.
exclusion	

Gao et al. (2016)

Reason for	The study was excluded because the results were presented in odds ratios and
exclusion	standardized mean differences could not be computed in the meta analysis

Hardcastle (2012)

Reason for	Exclude because it was prospective study. It was also excluded because it lacked
exclusion	a control group

Harland et al. (1999)

Reason for	The outcomes were assessed based on frequency and percentages of participants
exclusion	who had improved, making the outcome dichotomous hence excluded.

Harrison, Roberts & Elton (2004)

Reason for	The outcome was reported as odds ratio and couldn't be included in the meta
exclusion	analysis

Jimmy & Martin (2005)	
Reason for	The study is missing results necessary for conducting a meta analysis
exclusion	

Kastarinen et al. (2002)

Reason for	It was a randomised control trial which was assessing whether lifestyle
exclusion	counselling is effective in non-pharmacological treatment of hypertension in
	primary health care, hence could not answer the study objective

Kriska et al. (1986)

Reason for	The intervention was not primary care based
exclusion	

Long et al. (1996)

Reason for	The study did not have outcomes required in the intervention.
exclusion	

Marcus et al. (2007)

Reason for	The trial did not have a physician delivered intervention
exclusion	

Nymberg et al. (2018)

Reason for	The study was assessing the effect of mindfulness on adherence to physical activity
exclusion	on prescription

Pekmezi et al. (2016)

Reason for	The trial did not have a physician delivered intervention
exclusion	

Pinto et al. (1998)

Reason for	The study was excluded because the outcomes were different from the outcomes
exclusion	required in the review

Rome, Ekdahl & Gard (2009)

Reason for	The study was excluded because its primary outcome was direct and indirect costs
exclusion	of inactivity.

Schröder et al. (2018)

Reason for	The intervention was not health provider based.
exclusion	

Skår et al. (2011)

Reason for	This study examined the efficacy of two types of planning interventions (action
exclusion	plans and coping plans) in increasing physical activity levels when they are
	delivered via the Internet and not by a health care provider in primary care setting

Spittaels, Bourdeaudhuij & Vandelanotte (2007)

Reason for	The study was excluded because web-site-delivered physical activity intervention
exclusion	not involving a health provider or a primary care setting

Steele, Mummery & Dwyer (2009)

Reason for	The	study	was	excluded	because	the	face	to	face	or	Internet	delivered
exclusion	inter	ventior	ns wer	e not admi	inistered f	rom	a prir	nar	y care	sett	ing	

Stralen et al. (2009)

Reason for	Two tailored physical activity interventions, consisting of three tailored letters
exclusion	delivered during 4 months, were systematically developed by not administered
	from a primary care setting

Wanner et al. (2009)

Reason for	The study employed an Internet approach in providing advice and not a health
exclusion	provider given intervention

William et al. (2006)

Reason for	The interventions were not administered by a health provider or from a primary
exclusion	care setting

c. Characteristics of studies awaiting classification

A. Characteristics of ongoing studies

1. Missud et al. (2019)

Study name	Physical activity prescription for general practice patients with cardiovascular risk
	factors-the PEPPER randomised controlled trial protocol
Methods	The PEPPER clinical study is a randomised controlled trial to evaluate the efficacy
	over a period of 12months
Participants	35 to 74-year old patients with cardiovascular risk factors. 140 patients will be
	recruited in 15 GP practices and randomised in the intervention group or in the
	control group where patients will receive verbal advice of physical exercise
Interventions	based on structured information delivery, a personalised written physical activity
	prescription
Outcomes	Primary outcomes include the total weekly energy expenditure, measured by an
	accelerometer in Metabolic Equivalent Task-minute (MET-min). Secondary
	outcomes will include Clinical and morphological measures
Starting date	Results were expected at the beginning of 2019.
Contact	Correspondence: dmissud@gmail.com; laurent.connan@univ-angers.fr
information	
Notes	

2. Rose et al. (2007)		
Study name	The 'Women's Lifestyle Study', 2-year randomised controlled trial of physical activity	
	counselling in primary health care: rationale and study design	
Methods	The Women's Lifestyle Study is a two-year randomised controlled trial	
Participants	Participants included women aged 40–74 years undertaking less than 150 minutes of	
	at least moderate intensity physical activity per week recruited from primary care.	
Interventions	This study used an enhanced green prescription (referred to in this trial as a 'Lifestyle	
	script') that included telephone support from an RST exercise specialist over a nine-	
	month period, and involved a face-to-face visit with the primary care nurse at six	
	months to monitor progress and to provide additional support	
Outcomes	Study measures were assessed at baseline, 12 and 24 months. The primary outcome	
	measure was physical activity as assessed by the New Zealand physical activity	
	questionnaire long form	
Starting date	The results of the study were missing. Its unclear on the progress of dates	
Contact	Email: Sally B Rose* - sally.rose@otago.ac.nz;	
information	Beverley A Lawton - bev.lawton@otago.ac.nz; C Raina Elley - c.elley@auckland.ac.nz;	
	Anthony C Dowell - tony.dowell@otago.ac.nz; Anna J Fenton -	
	anna.fenton@oxfordclinic.co.nz	
Notes		

2. Rose et al. (2007)

Appendices

1. COCHRANE CENTRAL Search strategy 2020

#1MeSH descriptor: [Physical Fitness] this term only

#2MeSH descriptor: [Physical Exertion] this term only

#3MeSH descriptor: [Physical Education and Training] explode all trees

#4MeSH descriptor: [Sports] explode all trees

#5MeSH descriptor: [Dancing] this term only

#6MeSH descriptor: [Exercise Therapy] explode all trees

#7physical* near activ*

#8physical* near train*

#9physical* near fit*

#10exercise* near train*

#11exercise* near activ*

#12exercise* near physical*

#13sport*

#14walk*

#15bicycle*

#16exercise* near aerobic*

#17((life next style*) near activ*)

#18life-style* near activ*

#19lifestyle* near activ*

#20((life next style*) near physical*)

#21life-style* near physical*

#22lifestyle* near physical*

#23#1 or #2 or #3 or #4 or #5 or #6 or #7 or #8 or #9 or #10 or #11 #24#12 or #13 or #14 or #15 or #16 or #17 or #18 or #19 or #20 or #21 or #22 #25#23 or #24 #26MeSH descriptor: [Health Education] this term only #27MeSH descriptor: [Patient Education as Topic] this term only #28MeSH descriptor: [Primary Prevention] this term only #29MeSH descriptor: [Health Promotion] explode all trees #30MeSH descriptor: [Behavior Therapy] this term only #31MeSH descriptor: [Cognitive Therapy] this term only #32MeSH descriptor: [Primary Health Care] this term only #33MeSH descriptor: [Workplace] this term only #34promot* #35educat* #36program* #37#26 or #27 or #28 or #29 or #30 or #31 or #32 or #33 or #34 or #35 or #36 #38#25 and #37

2. MEDLINE Ovid search strategy

- 1. Physical Exertion/
- 2. Physical Fitness/
- 3. exp "Physical Education and Training"/
- 4. exp Sports/
- 5. Dancing/
- 6. exp Exercise Therapy/
- 7. exp Exercise/
- 8. (physical\$ adj5 (fit\$ or train\$ or activ\$ or endur\$ or exertion\$)).tw.
- 9. (exercis\$ adj5 (train\$ or physical\$ or activ\$)).tw.
- 10. sport\$.tw.
- 11. walk\$.tw.
- 12. bicycle\$.tw.
- 13. ((exercise\$ adj3 aerobic\$) or aerobics).tw.
- 14. ((lifestyle or life-style) adj5 activ\$).tw.
- 15. ((lifestyle or life-style) adj5 physical\$).tw.
- 16. or/1-15
- 17. Health Education/
- 18. Patient Education as Topic/
- 19. Primary Prevention/
- 20. exp Health Promotion/
- 21. Behavior Therapy/
- 22. Cognitive Therapy/
- 23. Primary Health Care/
- 24. Workplace/
- 25. promot\$.tw.
- 26. educat\$.tw.
- 27. program\$.tw.
- 28. or/17-27
- 29. 16 and 28
- 30. randomised controlled trial.pt.
- 31. controlled clinical trial.pt.
- 32. randomised.ab.
- 33. placebo.ab.
- 34. drug therapy.fs.
- 35. randomly.ab.
- 36. trial.ab.
- 37. groups.ab.
- 38. 30 or 31 or 32 or 33 or 34 or 35 or 36 or 37
- 39. exp animals/ not humans.sh.
- 40. 38 not 39
- 41. 29 and 40
- 42. (200412\$ or 2005\$ or 2006\$ or 2007\$ or 2008\$ or 2009\$ or 2010\$ or 2011\$ or 2012\$).ed.
- 43. 41 and 42

3. EMBASE Ovid search strategy

- 1. exp exercise/
- 2. fitness/
- 3. physical education/
- 4. exp sport/
- 5. dancing/
- 6. exp kinesiotherapy/
- 7. (physical* adj5 (fit* or train* or activ* or endur* or exert*)).tw.
- 8. (exercis* adj5 (train* or physical* or activ*)).tw.
- 9. sport*.tw.
- 10. walk*.tw.
- 11. ((exercise* adj aerobic*) or aerobic*).tw.
- 12. ((lifestyle or life-style) adj5 activ*).tw.
- 13. bicycle*.tw.
- 14. ((lifestyle or life-style) adj5 physical*).tw.
- 15. or/1-14
- 16. health education/
- 17. patient education/
- 18. primary prevention/
- 19. health promotion/
- 20. behavior therapy/
- 21. cognitive therapy/

- 22. primary health care/
- 23. workplace/
- 24. promot*.tw.
- 25. educat*.tw.
- 26. program*.tw.
- 27. or/16-26
- 28. 15 and 27
- 29. random\$.tw.
- 30. factorial\$.tw.
- 31. crossover\$.tw.
- 32. cross over\$.tw.
- 33. cross-over\$.tw.
- 34. placebo\$.tw.
- 35. (doubl\$ adj blind\$).tw.
- 36. (singl\$ adj blind\$).tw.
- 37. assign\$.tw.
- 38. allocat\$.tw.
- 39. volunteer\$.tw.
- 40. crossover procedure/
- 41. double blind procedure/
- 42. randomised controlled trial/
- 43. single blind procedure/
- 44. 29 or 30 or 31 or 32 or 33 or 34 or 35 or 36 or 37 or 38 or 39 or 40 or 41 or 42 or 43
- 45. (animal/ or nonhuman/) not human/
- 46. 44 not 45
- 47. 28 and 46
- 48. (200412* or 2005* or 2006* or 2007* or 2008* or 2009* or 2010* or 2011* or 2012*).dd.
- 49. 47 and 48
- 50. limit 49 to Embase

4. Web of Science search strategy

20 #19 AND #18

19 TS=(random* or blind* or allocat* or assign* or trial* or placebo* or crossover* or cross-over*)

- # 18 #17 AND #8
- # 17 #16 OR #15 OR #14 OR #13 OR #12 OR #11 OR #10 OR #9
- # 16 TI= (promot* or educat* or program*)
- # 15 TS=(workplace)
- # 14 TS= (primary health care)
- # 13 TS= (cognitive therap*)
- # 12 TS= ((behaviour or behavior) NEAR/2 therap*)
- # 11 TS= (health NEAR/2 promot*)

10 TS= (primary prevent*) # 9 TS= ((health educat*) or (patient* educat*)) # 8 #7 OR #6 OR #5 OR #4 OR #3 OR #2 OR #1 # 7 TS= ((lifestyle* or life-style*) NEAR/5 (activ* or physcial*)) # 6 TS= ((exercis* NEAR/2 aerobic*) or aerobic*) # 5 TS= (sport* or danc* or walk* or bicycle*) # 4 TS= (physical* educat*) # 3 TS= (exercis* NEAR/5 (train* or physical* or activ*)) # 2 TS= (physical NEAR/5 (fit* or train* or activ* or endur* or exert*)) # 1 TS= (exercis* therap*) 5. CINAHL Plus with Full Text EBSCO search strategy S34 S33 Limiters - Exclude MEDLINE records S33 S31 and S32 S32 EM 20041201-20121010 S31 S20 and S30 S30 S21 or S22 or S23 or S24 or S25 or S26 or S27 or S28 or S29 S29 TX allocat* S28 TX control* S27 TX assign* S26 TX placebo* S25 (MH "Placebos") S24 TX random* S23 TX (clinic* N1 trial?) S22 PT clinical trial S21 (MH "Clinical Trials+") S20 S10 and S19 S19 S11 or S12 or S13 or S14 or S15 or S16 or S17 or S18 S18 (TI promot* or educat* or program*) or (AB promot* or educat* or program*) S17 (MH "Work Environment") S16 (MH "Primary Health Care") S15 (MH "Behavior Therapy+") S14 (MH "Health Promotion") S13 (MH "Preventive Health Care") S12 (MH "Patient Education") S11 (MH "Health Education") S10 S1 or S2 or S3 or S4 or S5 or S6 or S7 or S8 or S9 S9 (TI sport* or walk* or bicycle* or exercis* or aerobic*) or (AB sport* or walk* or bicycle* or exercis* or aerobic*) S8 (TI physical N5 (fit* or train* or activ* or endur* or exert*)) or (AB physcial* N5 (fit* or train* or activ* or endur* or exert*))

S7 (TI exercis* N5 (train* or physical* or activ*)) or (AB exercis* N5 (train* or physical* or activ*))

S6 (MH "Exercise+") or (MH "Therapeutic Exercise+")

S5 (TI (lifestyle* or life-style*) N5 (activ* or physical*)) or (AB (lifestyle* or life-style*) N5 (activ* or physical*))

S4 (MH "Sports+") or (MH "Dancing+")

S3 (MH "Physical Education and Training")

S2 (MH "Physical Fitness")

S1 (MH "Exertion")

6. SPORTDISCUS search strategy

1.'physical activity'

2.exercise

3.fitness

4.sedentary

5.housebound

6.aerobics or circuits or swimming or aqua or jogging or running or cycling or fitness or yoga or walking or sport

7.patient education

8.primary prevention

9.health promotion

10.behaviour therapy

11.cognitive therapy

12.primary health care

13.workplace

14.controlled

15.randomised

16.random-assignment

17.double-blind

18.single-blind

19.clinical

20.placebos

21.comparative

22.evaluation

23.study

Creative Commons licensing terms

Authors will retain the copyright of their published articles agreeing that a Creative Commons Attribution 4.0 International License (CC BY 4.0) terms will be applied to their work. Under the terms of this license, no permission is required from the author(s) or publisher for members of the community to copy, distribute, transmit or adapt the article content, providing a proper, prominent and unambiguous attribution to the authors in a manner that makes clear that the materials are being reused under permission of a Creative Commons License. Views, opinions and conclusions expressed in this research article are views, opinions and conclusions of the author(s). Open Access Publishing Group and European Journal of Physical Education and Sport Science shall not be responsible or answerable for any loss, damage or liability caused in relation to/arising out of conflict of interests, copyright violations and inappropriate or inaccurate use of any kind content related or integrated on the research work. All the published works are meeting the Open Access under a Creative Commons attribution 4.0 International License (CC BY 4.0).